

Consolidated Tribal Health Project 6991 N. State Street Redwood Valley, CA 95470

P 707.485.5115 www.cthp.org

Request for Qualifications for:

Design-Builder Services
Issued by:

CONSOLIDATED TRIBAL HEALTH PROJECT
6991 N. STATE STREET
REDWOOD VALLEY, CA 95470
(707) 485-5115

Date issued: June 10, 2024

Due date:

Proposals must be submitted No later than 4:00 PM On July 19, 2024



Consolidated Tribal Health Project 6991 N. State Street Redwood Valley, CA 95470

Request for Qualifications (RFQ)

SECTION 1.0	Project Description2
	Space Requirements
SECTION 2.0	Purpose
SECTION 3.0	Notice to Submit a Statement of Qualification2
SECTION 4.0	Scope of Services3
SECTION 5.0	Project Delivery and Schedule Summary5
	Project Deliver
	Project Schedule Summary
SECTION 6.0	Proposal Content and Scoring6
	General Information
	Statement of Qualifications
	Demonstrate History of Successful Projects
	Budget Adherence
	Safety
	Past Utilization of TERO Certified Businesses
	Design-Builders Insurance Requirements
	Liquidated Damages
	Content
	Evaluation and Scoring
SECTION 7.0	Submission Requirements
	Point of Contact
	Questions Regarding this RFQ
SECTION 8.0	Attachments14
	Attachment 1 Consolidated Tribal Health Project – Concept Designs
	Attachment 2 DBIA Contract Document #530
	Attachment 3 DBIA Contract Document #535

#### 1.0 Introduction

Consolidated Tribal Health Project (CTHP) [Owner], a consortium of eight Tribes and Rancherias located in Mendocino County, CA, operates a comprehensive Tribal health clinic under a 93-638 compact with the Indian Health Service, including primary care, dental, and behavioral health care. The clinics operate out of a 1970s era campus which it acquired in the late 1980s. In 2023, CTHP's health clinic served an unduplicated population of nearly 5,000, and provided just over 20,000 visits. There is a significant opportunity to expand services, and improve health outcomes, through the expansion of the existing clinic, approximately 77% of whom are American Indians or Alaska Natives.

The proposed expansion project will add approximately 14,100 square feet of clinical space to the campus, together with roof top space. The additional space will consist of up to 16 dental operatories, 10 behavioral health rooms, two group therapy rooms, two waiting rooms, office space, meeting rooms, and storage. There will be some exterior work such as parking area expansion and site improvements.

# 2.0 Purpose

Consolidated Tribal Health Project Request for Qualification (RFQ) to solicit Statements of Qualifications (SOQ's) from General Contractors interested in providing Design-Build Services for the proposed Consolidated Tribal Health Project Expansion Project.

#### 3.0 Notice to Submit a Statement of Qualifications

Notice is hereby given that Consolidated Tribal Health Project has issued the solicitation for Design-Build Services for the proposed Consolidated Tribal Health Project Expansion Project.

RFQ documents will be provided with this solicitation. If unable to access documents, please contact the RFQ Coordinator. Other than access to documents, no phone calls and/or verbal request/inquiries about the content of this RFQ will be accepted. Individuals or Proposers who solicit information about this RFQ either directly or indirectly from other sources, other than obtaining copies of the RFQ, will be disqualified.

#### 4.0 Scope of Services

- 4.1 The Project has been approved by Consolidated Tribal Health Project Board of Directors
- 4.2 The proposed Guaranteed Maximum Price (GMP) for the project is approximately \$17,750,000.
- 4.3 Schedule. Anticipated Notice to Proceed: January 1, 2025.
  - Owner Occupancy is anticipated May 4, 2026.
- 4.4 Procurement Approach: Design Build (DB) delivery method for this project.
- 4.5 Consolidated Tribal Health Project plans to select the most qualified team based on criteria described in the RFQ.
- 4.6 A Price Proposal will be required including the following elements:
  - A. Price to include Overhead and Profit Fee Percentages.
  - B. Indian preference applies to this Project. Ten percent of the total number of available rating points will be given to qualified certified firms.
- 4.7 After Design-Builder is under contract, and as the design/budgets progresses, the Design-Builder will collaborate with and update Consolidated Tribal Health Project regarding the development of the Basis of Design Documents.
- 4.8 The Design-Builder will produce design documents for the Concept, Schematic, Design Development, 50% Construction Documents/Permit Set, 100% Construction Documents. The design documents shall be consistent with RFQ Section 4.0 Project Delivery/4.2 Project Schedule Outline duration.

Including providing a detailed estimate using CSI Master Format at each design milestone.

- Allow a minimum of two weeks for CTHP Team to review after each design milestone.
- 4.9 The Design-Builder shall review, analyze and validate the Owner Provided Information. Design Builder shall conduct such site investigations, environmental assessments, review of regulatory and legal authority and restrictions, and all other actions and review and assess other information as reasonably necessary to verify and validate the Owner provided information.

- 4.10 The Design Builder shall engage and work collaboratively with the Owner and designated Project Work Groups to progress the design to a sufficient state to develop the Basis of Design, the Guaranteed Maximum Price, and Project Schedule. The timing of the GMP Proposal and the percentage complete of the designs and specifications will be jointly determined by the Owner and the Design Builder.
- 4.11 Development of the GMP: Forecasting and development of accurate project cost estimates throughout each phase of the Project is vital to the Owner's financial management and strategy. The Owner relies on the Design-Builder to provide and validate current and detailed cost estimates and forecasts that will be incorporated into the overall cost controls for the Owner.
- 4.12 Throughout the Project, the Design-Builder will update estimates and forecasts and provide data to the Owner to reflect real time information. Design-Builder will provide all pricing, estimates, and other data to develop the Commercial Terms on an open and transparent basis. The project controls system used by the Design-Builder shall be acceptable to the Owner and will be capable of being broken down and reported in a number of different work breakdown structures, including but not limited to organizing the financial data by cost element codes, subcontracts, vendors, Construction Documents packages, etc.
- 4.13 Design-Builder will attend weekly meetings with project stakeholders throughout the design and construction phase.
- 4.14 Be fully responsible for the care, custody, and control of the project during the construction phase.
- 4.15 Provide all general conditions, general requirements and temporary facilities necessary to execute the project in an active healthcare facility.
- 4.16 Design-Builder shall provide, testing, and closeout of the Project pursuant to the Contract Documents. CTHP and the Design-Builder will discuss prior to setting the GMP if commissioning will be provided by a 3<sup>rd</sup> party contracted to CTHP or provided by the Design-Builder.
- 4.17 Federal Procurement Guidelines do apply. Davis-Bacon and Related Acts Regulations Apply.
- 4.18 Be fully responsible for the project safety during the construction phase.
- 4.19 Project Goals:
  - Build a trusting relationship with an honest project team that is committed to the Project, and is committed to exceeding client expectations by performing to a high standard of care, resulting in a quality-built Project that will last for future generations.
  - Through the use of local/regional trade partners, create an efficient facility that can be maintained with local resources, and provide a long service life.
  - Expand upon the expertise of the entire Project Team to enable the development of future

# REQUEST FOR QUALIFICATIONS (RFQ) projects.

 Achieve completion of the Project within the time frame and budget while providing scope and quality that exceeds CTHP expectations, while also maintaining a safe and healthy work environment.

### 5.0 Project Delivery and Project Schedule Summary

#### 5.1 Project Delivery

- 5.1.1 Use of the Design-Build delivery method provides an economic benefit by providing cost effective design and construction. CTHP is also using the Design-Build delivery method to contract with a single entity for both design and construction. The Design and Construction Services shall be set forth in a Design-Build Agreement. A draft form of the Design-Build Agreement is attached to this RFQ; however, it has not been vetted by Counsel and therefore CTHP reserves the right to make changes to the attachment Design-Build Agreement. Nevertheless, we anticipate using the DBIA 530 Form of the Contract as well as the DBIA 535 General Conditions, and Supplementary Conditions.
  - A. The Design-Builder Agreement is anticipated to follow the format below:
    - 1) The Design-Builder will be compensated on the basis of the Cost of the Work plus the Overhead and Profit Percentage Fee Proposal. The Design-Builder compensation will be subject to the Not to Exceed Amount as well as an established Guaranteed Maximum Price (GMP) that will be set at or below the Design-Build budget.
    - 2) The Design-Builder will have several phases. A general description of those phases are below:
  - B. Phase 1: During Phase 1, the Design-Builder will, among other duties, perform the following tasks:
    - 1) Complete the validation of all information provided by CTHP as well as the site investigation.
    - 2) Generate the deliverable set forth in the Phase 1 Scope of Services. Such deliverables are anticipated to include but not limited to: the Final Basis of Design Documents, the Final Schedule, and the fully developed GMP estimate budget, all of which shall be consistent with CTHP established GMP.
  - C. Phase 2: Phase 2 commences upon CTHP acceptance of the items submitted at the end of Phase 1, which form the basis of the GMP Agreement. In Phase 2, the Design-Builder completes final construction documents, secures all necessary permits, and completes construction pursuant to the Final Basis of Design Documents, the GMP and the Project Schedule.

#### **5.2 PROJECT SCHEDULE SUMMARY**

A preliminary project budget and schedule have been developed and reviewed internally by the client. A brief summary of current project schedule is as follows.

1. Commencement of Design-Builder Services:

2. Design and Construction:

3. Operations DAY 1:

Anticipated January 1, 2025 January 1, 2025 – June 1, 2026

May 4, 2026

#### PROJECT SCHEDULE OUTLINE

Program Confirmation
 Schematic Design
 Design Development
 Construction Documents
 Permit
 Pricing & Bidding
 Construction
 Weeks
 Weeks
 Oweeks
 Oweeks
 Oweeks
 Oweeks
 Oweeks
 Oweeks
 Oweeks
 Oweeks

Please note that the proposed timeline for the RFP process outlined herein is subject to change at the discretion of Consolidated Tribal Health Project. While we endeavor to adhere to the suggested schedule, unforeseen circumstances or adjustments to our internal procedures may necessitate alterations. We appreciate your understanding and flexibility in this matter.

#### 6.0 Proposal Content and Scoring

#### 6.1 GENERAL INFORMATION

The Statement of Qualifications shall demonstrate the Proposers ability to undertake the Project by providing the following technical and management qualifications of the Proposer, Team Members, and individual Key Team Members. The Proposer is responsible for ensuring that contract information contained in their referenced project profiles is correct. The inability to contact a reference may have a detrimental impact on the evaluating qualifications.

Emphasis will be placed on experience and expertise in performing substantive work on projects that are of Similar Scope and Complexity, as described in the definitions above. CTHP reserves the right to award more points to projects that have more of the characteristics in the definition of Projects of Similar Scope and Complexity. CTHP also reserves the right to award more points to successful projects in which Proposer, Team Members, and/or individual Key Members had substantial responsibility for their respective scopes of work.

#### 6.2 Statement of Qualifications – Proposer Organization and Responsibilities

A. Provide an organization chart (showing Team Members, Key Team Members and their firm affiliation) for all phases of the Project from programming through final acceptance and warranty and maintenance period. Key Team Members should include but not limited to the following individuals:

- a) Corporate executive(s) dedicated to the project.
- b) Design-Build Project Manager
- c) Construction Project Manager (if not the Design-Build Project Manager)

- d) Lead Estimator
- e) Construction Superintendent
- f) Design Manager
- g) Lead Designer
- h) Designer Project Manager
- i) Key Design Consultants

Clearly indicate whether any individuals are proposed to fill multiple roles. CTHP reserves the right to reject the inclusion of any individual or consultant firm from the winning Design-Builder.

- B. Provide a resume for all Key Team Members. Individual resumes should be no longer than one page and should include the following information:
  - 1) Description of Key Team Member's proposed Project role and the percentage of effort that the Key Team Member will contribute to the Project.
  - 2) Identification of Key Team Member's specialized experience and competence on Projects of Similar Scope and Complexity in the last 10 years.
  - 3) If applicable, name of college attended, dates of attendance, major course of study, and degrees/certification.
  - 4) If an architect or engineer, the states in which the architect or engineer is licensed to practice along with the applicable license numbers.
  - 5) Provide a brief statement of the Key Team Member's past experience and employment for the past 10 years.
  - 6) Design Build Institute of America (DBIA) designation, if applicable.
- C. Provide a narrative describing the qualifications of Proposer's Team Members and Key Team Members and why the Team proposed in this Statement of Qualifications will exceed CTHP's Project Goals.
- D. Provide litigation/dispute history for the lead Contractor and the Designer-of-record for the last five years.

#### 6.3 Demonstrate History of Successful Projects of Similar Scope and Complexity

- A. Describe the Team's experience in successfully managing design-build (or a similar integrated delivery model) Projects of Similar Scope and Complexity that include management and communications of an integrated team of design consultants, specialty subcontractors, and trade contractors. Include a description of any issues or problems that arose on the projects and how those issues of problems were resolved.
- B. It is not required that all Team Members have experience with Projects of Similar Scope and Complexity or with the Design-Build delivery method. However, Proposer should cumulatively possess experience in these areas such that CTHP has confidence in the capabilities of the Proposer as a whole.

#### 6.4 Budget Adherence

A. What process will the Proposer implement to ensure that the project is designed and constructed to a fixed fee and a set GMP? Include in your description projects where the Proposer creatively managed issues such as: sequencing, scheduling, site access, or other challenges factors. B. Describe the Proposer's approach to the following:

- Quality assurance/quality management
- Changes in Scope.
- C. What formal and informal protocols and processes will the Proposer implement to ensure a project that is "designed to the budget" the first time. Include the Proposer's experience in commissioning and testing Projects of Similar Scope and Complexity.
- D. Describe your project buyout process and how you progress from the preliminary budget to final construction budget.

#### 6.5 Safety

- A. Provide evidence of capacity of the Proposer to provide bonding in the amount of the estimated GMP. (An actual bond does not need to be submitted with Statement of Qualifications, but inability to provide the required bond capacity will result in disqualification).
- B. Within a three-year period immediately preceding the due date of the Statement of Qualifications, not have been determined by a final and binding citation and notice of assessment issued by department of labor and industries or through a civil judgement entered by a court of limited or general jurisdiction to have willfully violated, as defined in RCW 49.48.082, any provision of chapter 49, 46, 49.48, or 49.52 RCW.
- C. Provide the Table of Contents of the constructor's accident prevention program and a brief overview of its implementation.
- D. Provide the safety and accident prevention record of the construction members of the Proposer's team. Include other relevant information that documents their safety record, including Total Recordable Incidence Rate (TRIR)
- E. Provide a list of all OSHA, TOSHA, WISHA, or other safety agency citations and their dispositions for the past five years against the Proposer.

#### 6.6 Past Utilization of TERO Certified Businesses

Describe the Proposer's successful past utilization of businesses certified by TERO. An Indian-Owned businesses will be given preference in accordance with CTHP By-laws but non-Indian ownership does not preclude a business from submitting a proposal or being awarded the contract.

#### 6.7 Design-Builder Insurance Requirements

#### **6.7.1** Insurance Types and Limits

Design-Builder shall purchase and maintain insurance of the types, with limits of liability, containing such endorsements and subject to such terms and conditions as follows, as well as Article 5 of DBIA Document No. 535, Standard Form of General Conditions of Contract Between Owner and Design-Builder.

Design-Builder shall maintain the following insurance, naming the CTHP as an additional insured:

- a) Commercial General Liability Insurance in the amount of one million dollars each occurrence and three million dollars aggregate.
- b) Commercial Automobile Liability Insurance in an amount equal to the greater of one million dollars for all vehicles used in performance of the services or any other amount required by applicable law.
- c) Worker's Compensation Insurance, Disability Benefits Insurance and any insurance required by applicable law.
- d) No Subrogation. The Design-Builder waives all subrogation rights against CTHP and any of its Contractors, Subcontractors, Agents, Officers, Employees or Companies.
- e) CTHP Insurance. Design-Builder acknowledges that CTHP is covered under the Federal Tort Claims Act (FTCA) for tort claims for bodily injury and property damage as provided by federal law, and that CTHP acquires "gap" insurance only for tort claims caused by CTHP's negligence not covered by the FTCA. CTHP shall maintain commercially reasonable amounts of insurance or self-insurance covering its first party property damage exposure to damage to its buildings, facilities and business personal property.
- f) Professional Liability Insurance is to be provided by the Design Consultant. Such policies must provide coverage for the scope of professional services to be provided by or on behalf of the Design Consultant as set forth in the Agreement.
- g) Professional Liability Insurance is to be provided by Design-Builder. Such policies must provide coverage for the scope of professional services to be provided by or on behalf of the Design-Builder as set forth in the Agreement.
- h) Builder's Risk Insurance. Design-Builder shall procure and maintain from insurance companies authorized to do business in the state in which the Project is located builder's risk insurance on an "all risk" or equivalent policy form upon the entire Project to the full insurable value of the Project, including professional fees, overtime premiums and all other expenses incurred to replace or repair the insured property. The property insurance obtained by Design-Builder shall be the broadest coverage commercially available and shall include as additional insureds the interests of CTHP, Design-Builder, Design Consultants and Subcontractors of any tier. Such insurance shall include but not limited to the perils of fire and extended coverage, theft, vandalism, malicious mischief, collapse, flood, earthquake, debris removal and other perils or causes of loss as called for in the Construction Documents. A copy of the Builders Risk Insurance shall be made available to CTHP.
- i) Bonds and other Performance Security. Design-Builder to obtain performance and labor and material payment bonds, or other forms of performance security, the amount, form and other conditions of such security shall be as set forth in the Agreement. All bonds furnished by the Design-Builder shall be in a form satisfactory to CTHP.

#### 6.8 Liquidated Damages

Time is of the essence. CTHP and Design-Builder mutually agree that time is of the essence

with respect to the dates and times set forth in the Contract Documents.

Design-Builder understands that if Completion is not attained by the Scheduled Completion Date, CTHP will suffer damages which are difficult to determine and accurately specify. CTHP will levy a penalty of \$1,000/day against the Design Builder if the project is not completed by CTHP Completion date of May 1,2026. CTHP and Design-Builder will negotiate final liquidated damages with the GMP Amendment pursuant to the Agreement.

#### 6.9 Content

Proposals shall be no more than 25 pages. Proposer Statement of Qualifications must include a Table of Contents and be organized by discrete sections corresponding to the scoring criteria. A page is defined as an 8.5 x 11 inch (size when printed). For identification of a Project Table, Proposer may use 11x 17 inch (size when printed). Font size not to be smaller than 10 point. Statement of Qualifications not following the prescribed format will lose points. The following are excluded from the 25 page limit:

- Covers, tabs and dividers, provided that they do not contain substantive content
- Resumes
- Table of Contents
- Identification of Project Tables
- Cover Letter/Letter of Submittal

Provide the following information, in the order listed below:

- a. Title Page: State RFQ subject, full name of responding firm, mailing address, name and email address of contact person, and date of submission.
- b. Cover Letter: The cover letter should include statement indicating an understanding of the work to be performed and interest in performing the scope of work. Submittals in response to this RFP must be signed by the person in the Design-Builder organization who has the authority to commit the firm to the scope of work proposed in the submittal.
- c. Firm Background: Describe your firm including ownership structure, service area, volume, length of time in industry, financial stability and availability to the project locale. CTHP will give preference points to firms that are Native American owned, operated and controlled. If applicable, provide documentation to confirm your firm is at least 60% Native owned.
- d. Qualifications/Experience: Describe the Design-Builder experience in preparing the elements included in this RFP. Include at least three completed projects that are similar to the project described in this request. For each project, provide the following information:
  - Name and Location of project
  - Year completed
  - Name and contact information of client

- Name and contract information of client project manager
- Elements of the project that are similar to the scope elements of this project
- e. Approach: Provide a general discussion of your management philosophy. Include a description of your firm's involvement from the planning phase through the completion of the project. Provide specific examples of your methods to ensure quality, budget and schedule control utilizing inclusive, team-oriented processes.
- f. Staffing Plan: This section should contain a detailed and specific discussion of your firm's proposed staffing plan. Include an organization chart, description of staff roles, resume and other relevant biographical information as deemed necessary. Describe your plan to effectively provide services on several concurrent projects.
- g. Schedule and Cost Proposal: Provide a timeline showing the estimated length of time required for completion of the work described in the proposal. In addition, provide a detailed cost proposal that identifies all your associated costs and reimbursable for a not- to-exceed lump sum fee for this work. Also state any items not covered by your fee.

#### 6.10 Evaluation and Scoring

All proposals received by the July 19, 2024 deadline will be reviewed and evaluated by the selection committee. Proposals will be scored using the criteria below. The committee may review references and request interviews/presentations.

The selection committee's scoring will be tabulated and proposals ranked based on the numerical scores received. CTHP will be the sole judge for ranking proposals and determining which proposal best meets the selection criteria and CTHP's needs. CTHP is not required to explain or justify its decision process to bidders. CTHP may deem proposals non-responsive and remove them from consideration if proposals are submitted late, do not follow specified format, or do not include the requested information.

The evaluation criteria, weighting and maximum points, out of 100 are as follows:

- Firm Background: Evaluation of firm's ownership structure, size, service area, financial stability, and availability to meet the project needs. (0-10)
- Experience and Qualifications: Firm's experience in successful completion of projects of similar size and scope. (0-15)
- Project Approach: Proposed strategies to meet project objectives. (0-10)
- Staffing Plan: Ability of project manager and staff assigned to this project, with consideration given to project experience and staffing levels. (0-15)
- Schedule: Firm's availability to promptly begin project follow the execution of the contract and identify a realistic and expeditious project schedule (0-20)
- Costs: Cost, while a significant criterion, is not the only factor in the selection process. Cost is particularly important when all of the other evaluation criteria are relatively equal. (0-20)
- Preference points for Native American owned firms (10)
   A Native American owned firm must be able to provide proof that the Firm exercises majority control of the business and is substantially involved in day-to-day

management and operations.

#### 7.0 Submission Requirements

#### 7.1 Point of Contact:

Brian Sewell, Project Manager, Consolidated Tribal Health Project, is the point of contact (POC) for all matters concerning this RFQ. Questions regarding this RFQ and materials provided in response to this RFQ must be submitted to:

Brian Sewell – Mobile: 509-200-0588

brian@palousehillsconsulting.com

All communication between the Proposer and CTHP upon receipt of this RFQ shall be with the RFQ Point of Contact (POC).

Any other communication will be considered unofficial and non-binding on CTHP. Proposers are to rely on written statements by the RFQ POC through CTHP.

Communication direct to parties other than the RFQ POC will result in disqualification of the Proposer.

Any revisions to the RFQ, if it becomes necessary to revise any part of the RFQ, an addendum will be issued to the Proposers.

#### 7.2 Questions Regarding this RFQ:

Firm may submit questions regarding this RFQ prior to 4:00 pm PST on June 21, 2024. Only written questions submitted by email to <a href="mailto:lguzman@cthp.org">lguzman@cthp.org</a> will be accepted. No later than June 21, 2024, CTHP will make available by request a compilation of questions received (de-identify) and answers. Outside of this Q&A process, no additional information, other than what is included in the RFQ or is publicly available on CTHP's website, will be provided to potential bidders.

#### 7.3 Proposal Due Date and Estimated Timeline:

#### 7.3.1 Submission Deadline:

Proposals in response to this RFQ must be received no later than 4:00 pm on July 19,2024. Proposals to be emailed in pdf format to <a href="mailto:lguzman@cthp.org">lguzman@cthp.org</a>. Electronic Submittal Only. Submittals shall be limited to the documents specified in the RFQ document and shall not include additional brochures or other sales material that are not specifically requested in the RFQ.

It is strongly recommended that you follow up with a phone call to confirm your submittal has been received prior to the closing date and time. You may contact the RFQ Point of Contact at the contact information provided in section 7.1 of this RFQ.

#### 7.4 Timeline:

The deadlines and timelines included in the RFQ are subject to change by CTHP.

- June 21,2024 Deadline for consultants to submit and questions regarding RFQ.
- Junew 28, 2024 Answers to all RFQ questions distributed to all potential bidders.
- July 19, 2024 Proposals due to CTHP by 4:00 pm PST
- July 30 CTHP Selection Committee to review and score proposals and notify all respondents.
- Aug 16-19,2024 Interviews if deemed necessary by CTHP.
- Aug 23, 2024 CTHP Design-Builder selection and notification to all bidders.
- \*\*Site Walk/Project Site Visit has not been schedule, please contact Brian Sewells if Proposer will require to schedule a site visit. See Section 7.1 of this RFQ for Point of Contact Information.

#### 7.5 Selection Process

The basis of selection will be a scoring of proposals by the selection committee. The selection committee may invite recommended finalists for interviews based solely upon its evaluation of the selection criteria. The selection committee, at its sole discretion may forego the interview process. Selection of the successful firm will be entirely at the discretion of CTHP and CTHP reserves the right to waive minor irregularities in the selection process and to reject any and all proposals.

CTHP accepts no responsibility for the costs and expenses incurred by Proposer responding to this RFQ.

#### 7.6 Contract Requirements

Responding firms agree to comply with the legal requirements of the Consolidated Tribal Health Project, the standard and customary professional responsibilities for the Design-Builder Services as well as any special conditions which are made part of this solicitation or which are subsequently negotiated. Respondents also agree to comply with all applicable Federal, State, County and local ordinances, statues, rules and laws governing this project and its financing.

Respondents agrees that it will not sub-contract any part of the contract without the prior written consent of the Owner. The selection of all sub-consultants, engineers, or related services will be subject to prior approval from CTHP.

CTHP reserves the right to cancel award of a contract at any time before execution of the contract by both parties if cancellation is deemed to be in CTHP's best interest. In no event shall CTHP have any liability for the cancellation of the award.

Proposals that are incomplete or conditioned on the acceptance of additional terms by CTHP, or that contain any erasures, alterations, or that contain irregularities of any kind, or that are not in conformity with the law may be rejected.

#### 8.0 Attachments

Attachment 1 - Consolidated Tribal Health Project Concept Drawings\*

- Concept Design
- Overall Area Plan
- Floor Plan

**Attachment 2** - DBIA Contract Document #530 Standard Form of General Conditions of Contract Between Owner and Design-Builder – Cost Plus Fee with an Option for a Guaranteed Maximum Price

**Attachment 3** - DBIA Contract Document #535, Standard Form of General Conditions of Contract Between Owner and Design-Build.

\*The preliminary concept designs included in this RFP document are intended solely for illustrative purposes and do not represent the final design of the facility. Please be aware that these designs are subject to change, and the actual design, especially regarding interior design elements, may differ significantly. We encourage prospective bidders to consider these concepts as indicative of the project vision rather than definitive representations of the final outcome.

REQUEST FOR	QUALIFICATIONS	(RFQ
-------------	----------------	------

Attachment 1: Consolidated Tribal Health Project – Concept Designs

#### ATTACHMENT 2: DBIA Contract Document #530



STANDA DEC. JOF

AGREEML N. BET JEEN OWNER

AL DESIGN-L JILDER - COST

PLUS TE

TH AL OPTION FOR A

G' ARANTEED MAXIMUM PRICE

**Document No. 530** 

Second Edition 2010 © Design-Build Institute of America Washington, D



# Design-Build Institute of America - Contract Documents LICENSE AGREEMENT

By using the DBIA Contract Documents, you agree to and are bound by the terms of this License reement

- 1. License. The Design-Build Institute of America ("DBIA") provides DBIA Contract Documents and lice. their use a dwide. You acknowledge that DBIA Contract Documents are protected by the copyright laws of the United Society. You are a limited nonexclusive license to: (a) Use DBIA Contract Documents on any number of the contract purposes; and (c) copy DBIA Contract Documents into any machine-readable or printed form for backup or modification. Thousand the contract purposes in support of your permitted use.
- User Responsibility. You assume sole responsibility for the select fic doc nts or por is thereof to achieve your intended results, and for the installation, use, and results obt )BIA Cor. + Dr Jents. You acknowledge that ૂd from ti s and that consultation with an you understand that the text of the DBIA Contract Docume nas importa egal conseq attorney is recommended with respect to use or modifin of the t You will not represent that any of the contract documents you generate from DBIA Contract Documents a ે AIF ments unless (a) the document text is used without alteration or (b) all additions and changes to, and c' ns from, are clearly shown.
- 3. Copies. You may not use, copy, modify, or transfered portion, in whole or in part, except as expressly provided for in the license Reproad on of DBIA Contract Documents in printed or machine-readable format for resale or educational purposes is expressed by the produce and include DBIA's copyright notice on any printed or machine-readable conversed by modification, porturn reged into another document or program.
- 4. Transfers. You may not transfer post any copy, in diffication merged portion of DBIA Contract Documents to another party, except that a party with whom you recommend in the contract with you. You may not sublicense, sign, or feature like license except as expressly provided in this Agreement, and any attempt to do so is void.
- **5. Term.** The license is eff one year from date of purchase. DBIA may elect to terminate it earlier, by written notice to you, if you fail to comply the or condition this Agreement.
- nic files or other media by which DBIA Contract Documents are furnished to be free **6.** Limited Warranty. DBIA warrants the cannot be a carried with the cannot be a carried with the carried warranty. from defects in materials a formal use during the Term. There is no other warranty of any kind, expressed vorkm \_/nip ⊾. or implied, including, but not it .o the implied warranties of merchantability and fitness for a particular purpose. Some states ned warranties, so the above exclusion may not apply to you. This warranty gives you specific legal its and you may also I other rights which vary from state to state. DBIA does not warrant that the DBIA Contract Dod ients will med regui ents or that the operation of DBIA Contract Documents will be uninterrupted or error free.
- 7. Lim tions of Pares. A's earle liability and your exclusive remedy shall be: the replacement of any document not meeting DBIA Warranty' hich is returned to DBIA with a copy of your receipt, or at DBIA's election, your money will be refunded. In no event will slA be liable to you for any damages, including any lost profits, lost savings or other incidental or consequential damages and out of the use or inability to use DBIA Contract Documents even if DBIA has been advised of the possibility of ses, or for any claim by any other party. Some states do not allow the limitation or exclusion of liability for incidental or consequential damages, so the above limitation or exclusion may not apply to you.
- 8. Acknowledgement. You acknowledge that you have read this agreement, understand it and agree to be bound by its terms and conditions and that it will be governed by the laws of the District of Columbia. You further agree that it is the complete and exclusive statement of your agreement with DBIA which supersedes any proposal or prior agreement, oral or written, and any other communications between the parties relating to the subject matter of this agreement.

#### **INSTRUCTIONS**

For DBIA Document No. 530 Standard Form of Agreement Between Owner and Design-Builder - Cost Plus Fee with

an Option for a Guaranteed Maximum Price (2010 Edition)

#### Checklist

Use this Checklist to ensure that the Agreement is fully completed and all exhibits are attached.

	Page 1	Owner's name, address and form of business
	Page 1	Design-Builder's name, address and form of business Page
	1	Project name and address
	Section 2.1.3	Identify other exhibits to the Agreement Section
	4.2	Note the optional provisions that are provided
	Section 4.3.2	Complete blanks for additional sum for use of Work Product
	Section 5.2.1	Complete blanks for calendar days and note the optional language that wided
<del></del>	Section 5.2.2	Insert any interim milestones (optional)
	Section 5.4	Complete blanks for liquidated damages and no he optional provisions that provided
	Section 5.5	If the parties select the option provided they have to tan amount
	Section 5.6	Complete blanks for early completion note to tional provious that is provided
	Section 5.7	Note the optional provisions that provideo
	Section 6.1.2	Insert basis for pricing prelimin services (opti I)
	Section 6.2.1	Choose basis for Fee and co, te blanks
	Section 6.2.2	Insert financial arrangements to 'iustmer' and note optional provisions
	Section 6.3.3	Complete blanks for r 'rup; inser at a personnel names, etc.
	Section 6.3.4	Note the optional pro that is pro d
	Section 6.4.4	Note the optional provious † is provio
	Section 6.6.1.1 C	Complete blanks for GMP, dr. the optio. rovision that is provided Section
	6.6.1.2 Complete	e blanks for Design-Builder Conti. Cy
	Section 6.6.3.1 C	thoose many for sharing sungs; cunnete blanks Section
	6.7.1	Note wision
	Section 7.1.1	Complete hks is of me h
	Section 7.2.1	Complete by for rever centage and note optional provision
	Section 7.2.2	Note the optio. rovision to a provided
	Section 7.4	Complete blanks 1 +erest rate
	Section 8.1.	nethod for termination for convenience
	Section 8.2.1	Compie. 's for percentages
	Section 8.2.2	Comp' plan ercentages
	Section 9.1.1	ns Jwner's Senior Representative's name, etc. (optional)
	n 9.1.2	ert Owner's Representative's name, etc. (optional)
	Section 9.2.1	ert Design-Builder's Senior Representative's name, etc. (optional)
	Section 0 2 2	t Design-Builder's Representative's name, etc. (optional) Section 10.1
		At Insurance Exhibit
	.ion 10.2	Insert amount and conditions of bonds or other security and note the options that are provided
	Section 11.1	Insert any other provisions (optional)
	Inst Page	Owner's and Design-Builder's execution of the Agreement

# **General Instructions**

No.	Subject	Instruction
1.	Standard Forms	Standard form contracts have long served an important function in the United States and international construction markets. The common purpose of these forms is to provide an economical and convenient way for parties to contract for design and construction services. As standard forms gain acceptance and are used with increased frequency, parties are able to enter into contracts with greater certainty as to their rights and responsibilities
2.	DBIA Standard Form Contract Documents	Since its formation in 1993, the Design-Build Institute of America ("A") has regularly evaluated the needs of owners, design-builders, and other parties to the ("an-build proc" in preparation for developing its own contract forms. Consistent with ("an s mission ("an amulgating best design-build practices, DBIA believes that the design-build act show reflect a balanced approach to risk that considers the legitimate interests of all partice act shows design-build approach to the party best equipped to manage and minimize that risk, the goal promoting best design-build practices.
3.	Use of Non-DBIA Documents	To avoid inconsistencies among documents u. cor the same project, DBIA's Standard Form Contract Documents should not be used in conjunct with non-DBIA'd ments unless the non-DBIA documents are appropriately the according to the according to the project to ensure consistence.
4.	Legal Consequences	DBIA Standard Form Contra notume are legally binding contracts with important legal consequences. Cring paradvised and encouraged to seek legal counsel in completing or move these Dockings.
5.	Reproduction	DBIA hereby grants the purchase a limital cense to reproduce its Documents consistent with the License Agreemer. Cook wing the Documents. At least two original versions of the Agrant should be signed by the ortices. Any other reproduction of DBIA Documents is strictly property.
6.	difications	Effective conting is accomplished when the parties give specific thought to their contracting pals and then to the contract to meet the unique needs of the project and the design-build that rease these Documents may require modification for various purposes including, for expression provide the parties and laws, or to add special terms. DBIA's latest revisions to Docume. Provide the parties an opportunity to customize their contractual relationship selecting various optional contract clauses that may better reflect the unique needs and risks associated with the project.  Ye modifications to these Documents should be initialed by the parties. At no time should a coument be re-typed in its entirety. Re-creating the document violates copyright laws and destroys one of the advantages of standard forms-familiarity with the terms.
7.	Execu+;	It is good practice to execute two original copies of the Agreement. Only persons authorized to sign for the contracting parties may execute the Agreement.

# **Specific Instructions**

Section	Title	Instruction
General	Purpose of This Agreement	DBIA Document No. 530 ("Agreement") should be used when the parties intend that Owner pay Design-Builder the Cost of the Work plus a Fee, with or without a Guaranteed Maximum Price ("GMP"). If there is uncertainty about Owner's Project Criteria, or the Project Criteria remain to be developed by Owner and Design-Builder together, a cost-plus/GMP contracting approach is desirable.  If there is certainty as to Owner's Project Criteria, a lump sum fixe concept in the completion of all design and construction services may be suitable, especies when the Owner procures
Cananal	D	Design-Builder's services by competitive means. In such car and BIA Docume No. 525 should be used.
General	Purpose of These Instructions	These Instructions are not part of this Agreement, but are parties in their understanding of the Agreement and in completing the Agreen.
General	Related Documents	This Agreement shall be used in conjunction with the General Condition of Correct. Other related Contract Documents are listed in order of this Agreement.
General	Date	On Page 1, enter the date when both parties a final understanding. It is possible, due to logistical reasons, that the date to particle effect after of the greement will be the date recorded on Page 1. This ridges not, he wer, determine the terms article 5.
General	Parties: Owner and Design-Builder	On Page 1, ent legal no ration, partnership, limited partnership, limited liability company, or oth
2.1.2	GMP Exhibit, GMP Proposal	If a GMP is establis dup xecution his Agreement, the GMP Exhibit must be attached properties to Section 1.1. In MP is established after execution of this Agreement, the GMP exhibit and GMP exhibit must be attached properties.
2.1.5	Construction Docum	After exection of the Agreement, and consistent with the requirements of Section 2.4 of the General Contract, Design-Builder will prepare Construction Documents, subject of wheel's review and approval.
	Order of	The Cocuments are listed in Section 2.1 in the order of their precedence. The GMP whibit and MP Proposal are based on the Basis of Design Documents, which are comprised of various documents. The parties should strongly consider establishing the priority of the various documents comprising the GMP Exhibit or GMP Proposal to avoid disputes should discrepancies arise among the documents. Moreover, Section 2.1.3 recognizes that there may be other exhibits attached to this Agreement. If this is the case, the parties should discuss whether these exhibits should be part of the Basis of Design Documents. If these exhibits are not made part of the Basis of Design Documents, these exhibits will not take priority over the Basis of Design Documents in the event of a conflict.
3.3	ions	Terms, words and phrases used in the Agreement shall have the same meanings used in the General Conditions of Contract.
3.4	Design Specifications	The Owner is cautioned that if it includes design specifications in its Project Criteria there is case law holding that the Design-Builder is entitled to rely on such information, and to the extent such information is not accurate, the Design-Builder will be entitled to an adjustment in the Contract Price and/or Contract Time. Accordingly, the Owner to avoid such potential liability should consider using performance specifications.

Section	Title	Instruction
4.1	Work Product	This Agreement provides that the Design-Builder shall retain to yof the Work produces, but obligates Design-Builder to grant a limited license Owner to you we Work product according to the terms and circumstances described in Sect. 4.2, 4 of 4 and 4.5.
4.2	Owner's Limited License Upon Payment in Full	Design-Builder shall grant Owner, at Owner's risk, a limited license to use the Work Product at the completion of the Work in rection Owner's occur in of the Project. This Section also provides the partition of Sterring of Ship of some or all of the Work Product to the Owner on paymer full for a risk or ormed. Generally, where the Owner desires ownershop of Work Product it is sufficed transfer ownership of unique architectural and desig.
4.3	Owner's Limited License Upon Owner's Termination for Convenience or Design-Builder's Election to Terminate	Owner should rouse the min on for Convenience Clause to obtain Design-Builder's valuable design pots, and seek lower bids from other design-builders. Therefore, where Owner to have this Again ent for its convenience, and then decides to complete the Project with how thirdpart ces, Design-Builder shall grant Owner the rights set forth in Section 4.2 hrovid Owner particles. Design-Builder all amounts due Design-Builder as refered by the Connect Dodonnts, including paying Design-Builder an additional sum per control of the connection of the connecti
4.3.2	Additic Compensa	To minimize outes, the parties should negotiate prior to the execution of the Agreement the mount Owne. Il pay Design-Builder for the use of Design-Builder's Work Product in the enterminates this Agreement for its convenience or Design-Builder elects to inina agreement for cause. Enter this amount.
4.4	vner's Limitu License Upon Design-Builder's	If Design-Builder is properly terminated for default, Owner is granted a limited license to use the Work Product, to complete the Project, and Owner shall thereafter have the same rights and obligations as set forth in Section 4.2.
4.5	Owner's Indemnificat for Use of ' k	Owner's use or alteration of the Work Product shall be at its sole risk, and Owner must agree to defend, indemnify and hold harmless Design-Builder and anyone working by or through Design-Builder, including Design Consultants of any tier.
5.1	oate of Commencement	Design-Builder's obligation to commence work is triggered by its receipt of a Notice to Proceed unless the parties mutually agree otherwise.
5.2.1	Substantial Completion of the Entire Work	Enter the calendar days duration by which Substantial Completion has to be achieved. The parties in this Section have the option of modifying the definition of Substantial Completion set forth in the General Conditions of Contract if they want to use a Temporary Certificate of Occupancy as the benchmark. If this option is selected, Substantial Completion will be deemed to be achieved no later than the date a Temporary Certificate of Occupancy is issued if applicable to the Project.

Section	Title	Instruction
5.2.2	Interim Milestones	It may be that some portions of the Work must be completed by or within a provided period of time to accommodate Owner's needs. The parties may, eir option, it be not york. Enter the calendar days, starting from the provided by of Commencement, for the grade Substantial Completion of these identified portions of work. If these portions of the work are required to be substantially completed by certain makes the provided by the calendar days, attentions of the work are required to be substantially completed by certain makes, enter those dates. As presently drafted, no remedy is provided by the calendar days, starting from the calendar days, eir option, it is only these portions of the calendar days, starting from the calendar days, eir option, it is only these portions of the calendar days, starting from the calendar days, eir option, it is only these portions of the calendar days, eir option, it is option to the calendar days, eir option, it is option, it is option. The calendar days, eir option, it is option, it is option, it is option, it is option. The calendar days, eir option, it is option. The calendar days, eir option, it is option, it is option. The calendar days, eir option, it is option to the calendar days, eir option, it is option. The calendar days, eir option,
		Owner should relieve a good the uation of the amount that is reasonably necessary to compensate it tray. Owner all uld not establish liquidated damages to penalize Design-Builder. Moreov to event a tray is not established upon execution of the Agreement, it appears prudent to the tries to retain from establishing liquidated damages until such time as the GMP is established.
5.4	Liquida\ Damage.	establishe grace detected between the Scheduled Substantial Completion Date and assume of a idated damages in order to prevent disputes as to which party bears responsibly for the days of delay. The parties should enter the calendar days that may pass folling the School disputes as to which party bears responsibly the School days of delay. The parties should enter the calendar days that may pass folling the School dispute damages will be assessed.  The parties are to provided the option of establishing liquidated damages if the Designstalls to achieve Final Completion within a specified number of days after Substantial completion is selected, the parties have to negotiate the number of days, as well the liquidated damages amount. The parties in negotiating liquidated damages should keep in mind that the amount of liquidated damages for failing to achieve Final Completion should be a considerably scaled down amount and should reflect the financial harm to the Owner. In no case should the total amount of liquidated damages for the Project exceed an amount that is reasonably necessary to compensate Owner for Project delay.
		The parties also have the option here of eliminating liquidated damages altogether, in which case the Owner can recover actual damages for Project delay at an amount that is capped by the parties. The Owner is cautioned that it still cannot recover consequential damages, as they are waived under Section 10.5.1 of the General Conditions of Contract.
5.5	Liquidated Damages Cap	The parties can agree to cap liquidated damages for delay at a negotiated amount.

Section	Title	Instruction
5.6	Early Completion Bonus	If the Project economics justify liquidated damages, then it priate to councies liquidated damages with an early completion bonus. The parties and enter the imber of calendar days prior to the Scheduled Substantial Completion Date to a lill set the imber of calendar days prior to the Scheduled Substantial Completion Date to a lill set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the Scheduled Substantial Completion Date to all set the imber of calendar days prior to the set the imber of calendar days prior to the imber of calendar days
5.7	Compensation for Force Majeure Events	The parties are provided coport of providing the Design-Builder the right to receive compensation of the parties agree to modify Section 8.2.2 or the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate and the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have to negotiate how many cumulative of Fore the parties have
6.1.2	Optional Pricing	nement allo the p. the flexibility to establish within the Contract Price a a. ant, ant basi or certail preliminary portions of the Work which may be necessary to p. Desib. I'del furnish Owner with a GMP. Alternatively, the parties may use DBIA Docum. No. 520 c. rm certain preliminary design services prior to setting the GMP. Enter a de. tion of any such services, the basis for determining the price, and the price to be paid.
6.2.1	Design-Builo Fee	Er Ount of Design-Builder's Fee as a sum certain or as a percentage of the Cost of the Ork. Design Builder's Fee shall be commensurate with the services it provides and the risk it assumes in providing single point responsibility to Owner.
6 !	Adjustments to r's Fee	For additive Change Orders, the parties have to negotiate the Fee the Design-Builder will receive. For deductive Change Orders, the parties have the option by checking the appropriate box to signify whether there will be no additional reduction or whether there will be an additional reduction based on a negotiated percentage.
6.3.3	Wages for gn- Cuilder's Joyees Jicipal or Branch Offices	DBIA endorses reimbursing salaries and associated benefits of Design-Builder's Project personnel, such as accountants, stationed at offices other than the field office, when to do so is more efficient and cost effective. Enter the percentage markup to be applied for Project-related overhead associated with such personnel. Insert, or attach as an exhibit, a list of such personnel and their job functions.
6.3.4	Employee Benefits	It may be simpler for the parties to agree on a multiplier (rather than actual costs) to compensate the Design-Builder for employee benefits. Accordingly, the parties may want to insert the multiplier to be applied to the wages and salaries of such reimbursable employees.

Section	Title	Instruction
6.3.7	Costs for Defective/Non- Conforming Work	The Cost of the Work shall include the costs to repair or corrective or non-comining Work (including warranty or corrective work performed after Suntial Completed, unless caused by Design-Builder's negligence. DBIA believes that Designed ilderted and not be penalized for inadvertent mistakes which relinevitable when designed constructing a Project. To do so would encourage ultra-comvatism in every task, the unless cost of which would be greater than a proactive approach to forming the Work.
6.3.23	Warranty Escrow	At this section, the parties a provided appropriate operation of the warrant ariod, the analysis share any savin. We are the Design-Builder for its remain under appropriate operation of the warrant ariod, the analysis share any savin. We are the expression of the warrant ariod, the analysis share any savin. We are the expression of the warrant ariod, the analysis are returned to the Owner subject to Design-Builder's share any savin. We are the expression account is exhausted, if funds remain under appropriate to provide a provided a provided a provided and are the design-Builder for its warranty Work.
6.4.4	Allowance Value	This section recogn is the parties agree that certain items of Work should be treated as Allowance Item and probased on Allowance Values. The Allowance Value for which Builder with the parties and insurance associated with the Allowance Item. All on the State of the Allowance Item, such as design fees, general conditions costs and fee, deemed and included in the Contract Price. However, by checking the box, the parties agree at in the event the actual cost of the Allowance Item is greater than or less than the Allowance by a negotiated percentage, then Design-Builder's right to Fee and markup determines pursuant to Section 6.2.2.
6.0	e Guarantı Maximum P	This Agreement provides the parties flexibility in establishing the Contract Price. Parties can establish a GMP before or after entering into this Agreement, or elect to proceed on the basis of costs plus a fee, without a GMP.  If a GMP method is elected, the GMP should not be established until the Basis of Design Documents are sufficiently defined to make the GMP realistic and meaningful. Setting it too early does not permit reasonable opportunity for scope definition and evaluation of Project risk. On the other hand, setting it too late may not achieve Owner's objective of having an early price guarantee to enable it to make decisions relative to the Project.

Section	Title	Instruction
6.6.1.1	GMP at Agreement Execution	Enter the GMP, if appropriate. Attach as an exhibit to this
6.6.1.2	GMP Continancy	Enter the amount of Desig. ilder' intingency. The Contingency is for the exclusive use of the Design-Buil and covers incicipated costs incurred that are not the basis of a Change Order. This sect forth by of example only the type of costs that would be funded out of the Continum. There cost is also not limited to any deductibles the Design-Builder is obligate to provide t

Section	Title	Instruction
6.6.2.1	GMP Proposal After Execution of This Agreement	At the request of Owner, Design-Builder shall submit its GMP. which shall in a the items listed in Sections 6.6.2.1.1 to 6.5.2.1.9. If the parties agree diltions or delines from this list, modify this Section 6.6.2.1 appropriately.  The Agreement provides the parties with dibility as to when the Graph Proposal will be submitted after execution of the Agreement. To execution of the Agreement the parties should discuss when Owner design 19-Build Submit its Graph Proposal.
6.6.2.1.4	Schedule	Given that expedited delity is one of the rimary factory ing many owners to select the design-build method, it is strongly belity at that the party should discuss and understand what each party must a support it roject schedule. The entire Work, both design and construction, should be schooled indicate the dates for the start and completion of four stage and entire work, including the date when Owner information and approvals are in and any mer created constraints. The Agreement also provides flexibility to estate here is cheduled. Stantial Completion Date prior to submission of the GMP Proposal.
6.6.2.3	Acceptance of GMP Proposal	Group a Con ct Document pursuant to Section 2.1.2.
6.6.2.4	Failure to Acco	This Ab ment province options for Owner in the event it fails to accept the GMP Proposal a volchoices or Design-Builder if Owner fails to exercise any of the three options. These option is specifically designed to prevent one party from receiving a windfall in the the partie. Inot agree on the GMP and the Agreement is terminated.  a partie ould take note that if Owner exercises its option to terminate for convenience, or Design-Builder suspends performance, Design-Builder will not be entitled to payment for uncompleted Work provided by Section 8.2. However, additional payment for Owner's use of Work Product will be due Design-Builder pursuant to Section 4.3, if Owner proceeds to complete the Project using Design-Builder's Work Product.
6.6.3	Savings	One of the benefits of a GMP approach is the possibility that with good management by Design-Builder and timely support from Owner the actual Cost of the Work and Fee may be less than the GMP. This creates a savings pool that should result in a benefit to both Design-Builder and Owner. Sharing these savings creates an incentive for Design-Builder to save costs. Some factors to consider in determining how the Savings are shared include the timing for the establishment of the GMP and the amount of Design-Builder's Fee established under Section 6.2.1.
6.6.3.1	Savings Calculations	This section provides that if the actual Cost of the Work and Design-Builder's Fee is less than the GMP, as such GMP may have been adjusted, the savings, if any, shall be shared. The Agreement offers two choices for distributing Savings. Choose a method and enter the appropriate figures.

Section	Title	Instruction
6.7	Performance Incentives	In addition for the potential of the Design-Builder to share in the as set forth in a set forth in the continuous set forth in an extension of this Agreement. As the continuous set forth in an extension of this Agreement.
7.1.1	Progress Payments	Enter the day of the mo <sup>-</sup> when Des Builder s su <sup>+</sup> its Application for Payment.
7.2.1	Retainage	Enter the percentage Ow will retain om Progress Payments to Design-Builder until fifty percent (50%) the Work are neted. Owner should recognize that it creates undue hardship to he image on contractors that have completed their work early in the Project. Owner to according onsider releasing retainage on Subcontractors that complete work early in Project, iding that these Subcontractors have satisfactorily performed their potential of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage provision by checking the box. The image of the option of modifying the retainage of the option of modifying the retainage of the option of the option of modifying the retainage of the option of the option of modifying the retainage of the option of th
7.2.2	Release of Retainage	T' so guires the Owner to release retainage to the Design-Builder. If the Design-Builder and Owner move established a warranty reserve in accordance with Section 6.3.2.4, the parties shall establish an escrow account at this time.
		The parties should enter the rate at which interest will accrue on Design-Builder's payments if unpaid five (5) days after due. Late payment creates a hardship for Design-Builder, its Design Consultants and Subcontractors.
7.5	Record Keeping	The Owner is provided access to Design-Builder's accounting information as it relates to Costs of the Work. However, if the parties have agreed to multipliers or markups, the time to challenge and negotiate those percentages is at the time the parties execute the Agreement and not during the Project or after it has been completed. Accordingly, the Owner can at any time audit these percentages only to confirm that such percentage has been properly charged and not to challenge the composition of such percentage.

Section	Title	Instruction	
8.1.3	Termination for Convenience: Overhead and Profit	The parties should choose prior to execution of the Agreeme ethod that will used to determine overhead and profit paid to Design-Builder in the even when termin Design-Builder for its convenience. The parties may choose to set percentage ethead and profit prior to execution of the Agreemen may choose to determine the sums to be paid for overhead and profit at the time the termination. If the pass choose to set overhead and profit rates prior to execution the Agreement, the percentages should be entered in Section 8.1.3.	
8.2	Termination for Convenience: Additional Payments	Although it is important Owner to I : a proce. In minating this Agreement for convenience, the procenust consider of interests of a gn-Builder. If Owner terminates this Agreement for its own pensating Design-Builder for its costs will not be adequate becaute Design-Learn nave committed its resources for a small amount of revenue. There is addition to eleverhead and profit paid in Section 8.1, Owner shall pay Design-Builder and contract Price or, in the has not be in stablished, the remaining balance of the most recent estimated Contract Price. For the procentages Owner shall pay Design-Builder if Owner training this Agrement for interests of a proce. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process. In minating this Agrement for interests of a procent process of a procent process. In minating this Agreement for interests of a procent process. In minating this Agreement for interests of a procent process. In minating this Agreement for interests of a procent process. In minating this Agreement for interests of a procent process. In minating this Agreement for interests of a procent process of a procen	
8.3	Termination for Convenion Owner's ec Work Prost	Own, 'ould 'o Termination for Convenience clause to obtain Design- Builder's valuable fign conce, and then seek lower bids from another design- builder. If Owner terminates Agreement for its own convenience, and chooses to proceed with the Project rising Design Fer's Work Product, Owner should pay an additional sum for the use of Ruilder's Froduct pursuant to Section 4.3.	
Ar e	Renr sof	Owner's Representative at Sections 9.1.1 and 9.1.2, respectively.  Enter the name, title, address and telephone number of Design-Builder's Senior Representative and Design-Builder's Representative at Sections 9.2.1 and 9.2.2, respectively.  The parties can elect to establish these Representatives during the performance of the Project rather than at the time of execution of this Agreement. If Representatives are identified after execution of the Agreement, an appropriate amendment should be made to the Agreement at the time these individuals are designated.	
10.1	Insurance	Attach an Insurance Exhibit setting forth in detail the insurance coverages required for the Project. Parties are advised to familiarize themselves with the terms of Article 5 of the General Conditions of Contract, Insurance and Bonds, and to consult their insurance advisor.	
10.2	Bonds	Enter the type and amount of bonds or other performance security required for the Project. Where bonding is not required by statute, Owner may want to evaluate the project risks versus the bonding costs in deciding what type of performance security to require.	

Section	Title	Instruction	
11.1	Other Provisions	Insert any other provisions. For example, the parties may through litigation rather than arbitration in which case the optionshould be included.	language in Section



## Attachment 3 DBIA Contract Document #535

# TABLE OF CONTENTS

Article	Name	Page
Article 1	Scope of Work	2
Article 3	Interpretation and Intent	2
Article 4	Ownership of Work Product	3
Article 5	Contract Time	4
Article 6	Contract Price	
Article 7	Procedure for Payment	4
Article 8	Termination for Convenience	15
Article 9	Representatives of the Parties	16
Article 10	Bonds and Insurance	
Article 11	Other Provisions	18



# Standard Form of Agreement Between C ∠ner and Design-Builder - Cost Plus Fee with an Option for a Guaranteed Maximum is ce

This document has important legal consequences. Consultation with an attorney is recommended with respect to its completion or modification.

This <b>AGREEMENT</b> is made	de as of the	day of	
			th the Project
OWNER: (Name and address)			
<b>DESIGN-BUILDER:</b> (Name and address)			
PROJECT:			
(Inclur' ame and local	will appear in the Contract Documents)		
In consideration of the mu forth herein.	tual covenants and obligations contained hereir	, Owner and Design-Build	er agree as set

DBIA Document No. 530 Page 1

# Article 1

#### Scope of Work

**1.1** Design-Builder shall perform all design and construction services, and provide all material, equipment, tools and labor, necessary to complete the Work described in and reasonably inferable from the Contract Documents.

# Article 2 Contract Documents

- **2.1** The Contract Documents are comprised of the following:
  - **2.1.1** All written modifications, amendments, minor changes, and Change Orders this Agreent issued in accordance with DBIA Document No. 535, Standar' Form of General Conductor of Atract Between Owner and Design-Builder (2010 Edition) ("General Conductor");
  - **2.1.2** The GMP Exhibit referenced in Section 6.6.1.1 or, if a<sub>k</sub> able, the GMP posal accepted by Owner in accordance with Section 6.6.2 herein;
  - **2.1.3** This Agreement, including all exhibit, List for exame 2, performance incentive arrangements, markuperiority ibits, allow ces, unit prices, or exhibit detailing offsite reimbursable personnel) but excluding if application in Exhibit;
  - 2.1.4 The General Conditions of Con no and
  - **2.1.5** Construction Documents prepare and roved in cordance with Section 2.4 of the General Conditions of Contract.

## <u>· e 3</u>

#### i. roretation and Intent

- 3.1 Design-Build and Ownto execution of the Agreement (and again, if applicable, at the time of acceptance of the GM ropos by Own accordance with Section 6.6.2 hereof), shall carefully review all the Contract Documents, including the Proposal of the Agreement of the Agreement
- the Contract Documents within the Contract Time(s) for the Contract Price. The Contract Documents are intended to be commentary and interpreted in harmony so as to avoid conflict, with words and phrases interpreted in a montract period of the Contract Documents are discovered after execution of the Agreement, or if applicable, after Owner's acceptance of the GMP Proposal, Design-Builder and Owner shall attempt to resolve any ambiguity, conflict or inconsistency informally, recognizing that the Contract Documents shall take precedence in the order in which they are listed in Section 2.1 hereof. (Note, the parties are strongly encouraged to establish in the GMP Exhibit or GMP Proposal (as applicable) the priority of the various documents comprising such exhibit or proposal.)

- **3.3** Terms, words and phrases used in the Contract Documents, including this Agreement, shall have the meanings given them in the General Conditions of Contract.
- **3.4** If Owner's Project Criteria contain design specifications: (a) Design-Builder shall be entitled to reasonably rely on the accuracy of the information represented in such design specifications and their compatibility with other information set forth in Owner's Project Criteria, including any performance specifications; and (b) Design-Builder shall be entitled to an adjustment in the Contract Price and/or Contract Time(s) to the extent Design-Builder's cost and/or time of performance have been adversely impacted by such inaccurate design specification.
- 3.5 The Contract Documents form the entire agreement between Owner ar esign-Builde and by incorporation herein are as fully binding on the parties as if repeated herein. No correspond to or other agreements have been made by the parties except as specifically stated in the Contract rument

#### **Article 4**

#### Ownership of Work Produ

- 4.1 Work Product. All drawings, specifications and oth electronic d , including such nents . documents identified in the General Conditions of Contr પાilder Owner under this furnis by Des. Agreement ("Work Product") are deemed to be instrume of service ai Design-Bu. , retain the ownership and property interests therein, including but not lid to any int ctual propert, ights, copyrights and/or patents, subject to the provisions set forth in Sections 4. rough 4
- 4.2 Owner's Limited License upon Project nletion a ayment in Full to Design-Builder. Upon Owner's payment in full for all Work performed under to C. act Docu ts, Design-Builder shall grant Owner a limited license to use the Work Product in connection the ar's occupy of the Project, conditioned on Owner's express understanding that its alteration of the ork is ct without the involvement of Design-Builder is at Owner's sole risk and without liz' r legal expos to De - Builder or anyone working by or through Designw tier (c ectively le "Indemnified Parties"), and on the Owner's Builder, including Design Consu. corth ... tion 5 below. obligation to provide the indemnity

#### [A ` ` parties' option, ` of the following may be used in lieu of Section 4.2.]

Upon Owne pay. If Ill for all Very performed under the Contract Documents, Design-Builder: (a) grants Owner a limite icense to Work Froduct in connection with Owner's occupancy of the Project; and (b) transfers all owner of and opercy wests, including but not limited to any intellectual property rights, copyrights and/or pater is that portion of the Work Product that consists of architectural and other design elements as specification that are unique to the Project. The parties shall specifically designate those portions of Work Product for who ownership in the Work Product shall be transferred. Such grant and transfer are inditioned or the work product shall be transferred. Such grant and transfer are inditioned or the work product without the involvement of sign-Buil at the arise are indicated as a risk and without liability or legal exposure to Design-Builder or anyone working by Design-Buil and Design Consultants of any tier (collectively the "Indemnified Parties"), and on the Owner's obligation of provide the indemnity set forth in Section 4.5 below.

or

Upon Owner's payment in full for all Work performed under the Contract Documents, Design-Builder transfers to Owner all ownership and property interests, including but not limited to any intellectual property rights, copyrights and/or patents, in the Work Product. Such transfer is conditioned on Owner's express understanding that its alteration of the Work Product without the involvement of Design-Builder is at Owner's sole risk and without liability or legal exposure to Design-Builder or anyone working by or

through Design-Builder, including Design Consultants of any tier (collectively the "Indemnified Parties"), and on the Owner's obligations to provide the indemnity set forth in Section 4.5 below.

**4.3** Owner's Limited License upon Owner's Termination for Convenience or Design-Builder's Election to Terminate. If Owner terminates this Agreement for its convenience as set forth in Article 8 hereof, or if Design-Builder elects to terminate this Agreement in accordance with Section 11.4 of the General Conditions of Contract, Design-Builder shall, upon Owner's payment in full of the amounts due Design-Builder under the Contract Documents, grant Owner a limited license to use the Work Product to complete the Project and sequently occupy the Project, and Owner shall thereafter have the same rights as set forth in Section 4.2 about conditioned on the following:

<b>4.3.1</b> Use of the Work Product is at Owner's sole risk without liability legal ensure to any Indemnified Party, and on the Owner's obligation to provide the indemnity set for in scion 4.5 below and
4.3.2 Owner agrees to pay Design-Builder the additional sure of
<b>4.4 Owner's Limited License upon Design-Builder' efault.</b> If the greement annated due to Design-Builder's default pursuant to Section 11.2 of the Gene conditions of a tract, then Design-Builder grants Owner a limited license to use the Work Product to complete the piect are absequently occupy the Project, and Owner shall thereafter have the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates. So as set to be under the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates. So as set to be under the same rights and obligates as as set to be under the same rights and obligates. So as set to be under the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates. So as set to be under the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates. So as set to be under the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights are same rights and obligates as as set to be under the same rights and obligates as as set to be under the same rights and obligates are same rights and obligates as as set to be under the same rights are same rights. The same rights are same rights are same rights and obligates are same rights and obligates are same rights. The same rights are same rights. The same rights are same rights. The same rights are same righ
4.5 Owner's Indemnification is of Work roduct. Owner is required to indemnify any Indemnified Parties based on the use or alteration. Work Project under any of the circumstances identified in this Article 4, Owner shall defend, indemnify an old had so su Indemnified Parties from and against any and all claims, damages, liabilities, losses and expendincluding any including heys' fees, arising out of or resulting from the use or alteration of the Work roduct.
Article 5
Contract Time
pate of Commence with the Work shall commence within five (5) days of Design-Builder's receipt of wher's Notice of Commencement") unless the parties mutually agree otherwise in writing.
5. Substantial Completion and Final Completion.
stantial Completion of the entire Work shall be achieved no later than
[At the parties' option, the following supplemental language may be inserted at the end of Section 5.2.1 in the Project is subject to a Temporary Certificate of Occupancy.]
The parties agree that the definition for Substantial Completion set forth in Section 1.2.18 of the General Conditions of Contract is hereby modified to read as follows:

"Substantial Completion is the date on which the Work, or an agreed upon portion of the Work, is sufficiently complete in accordance with the Contract Documents so that Owner can occupy and use the Project or a portion thereof for its intended purposes, provided, however, that Substantial Completion shall be deemed to have been achieved no later than the date of issuance of a Temporary Certificate of Occupancy issued by the local building official."

- Interim milestones and/or Substantial Completion of identified portions of the York ("Scheduled 5.2.2 Interim Milestone Dates") shall be achieved as follows: (Insert any interim milestones for r different scheduled dates for Substantial Completion)
- 5.2.3 Final Completion of the Work or identified portions of the Work shall be achie as expedi\* as reasonably practicable. Final Completion is the date when all Work is complete pursual. **dition** of Final Completion set forth in Section 1.2.7 of the General Contract.
- All of the dates set forth in this Article 5 (collectionly the \*ract Time(s)") / "I be subject to adjustment in accordance with the General Condition
- 5.3 Time is of the Essence. Owner and Design-Buil , mutually ag that time is ie essence with respect to the dates and times set forth in the Contract Docun.

5.4 Liquidated Damages. Design-Builder restands Substantial Completion is not attained by the
Scheduled Substantial Completion Date, Owner 🕠 Ger dama which are difficult to determine and accurately
specify. Design-Builder agrees that if Substantial m, in is not ined by
() day. fterchedulebstantial Completion Date (the "LD Date"),
Design-Builder shall pay Owner Dollars (\$) as liquidated
damages for each day that Sub. Toletion each day the LD Date. (If a GMP is not established upon execution of this Agreement, the parties should consider the stable of the same of the sam
[The parties may want to con. the folio. Supplemental language within Section 5.4 if they want to assered amages, Failing to meet Final Completion. In this case, the first sentence in
5. 5 2.3 should Le leted and replaced with the following language.]
$\sim$ 52.3 should $\sim$ leted and replaced with the following language.]
Design-Buildenderstan Final Completion is not achieved within
Design-Builde nderstar Sinal Completion is not achieved within

(In lieu of th quidated damages specified in Section 5.4 or the alternate provided herein, the Parties may deci that the Agreement will provide for actual damages in the event of Project delay, with Owner ig cautioned that there is a waiver of consequential damages under Section 10.5.1 of the General Conditions of Contract. In this case, delete Sections 5.4 and 5.5 and insert the following.]

5.4 Design-Builder and Owner have agreed not to provide for liquidated damages in this Agreement for failure of Design-Builder to achieve the Contract Time(s) set forth in this Article 5. Design- Builder understands, however, that Owner may suffer actual damages in the event the Contract Time(s) set forth herein are not timely achieved. Owner shall be able to recover such actual damages from Design-Builder to the extent it can demonstrate that actual damages have been incurred, are directly related and caused by Design-Builder's failure to meet the Contract Time(s) set forth herein, and are not

5 1

waived by Section 10.5.1 of the General Conditions of Contract. Notwithstanding the foregoing, in no event shall Design-Builder's liability for actual damages for delays exceed
Dollars (\$).
5.5 Any liquidated damages assessed pursuant to this Agreement shall be in lieu of all liability for any and all extra costs, losses, expenses, claims, penalties and any other damages, whether special or consequential, and of whatsoever nature, incurred by Owner which are occasioned by any delay in achieving the Contract Time(s).
[The Parties may also desire to cap the liquidated damages payable under this Agreent, in which case the following language should be included at the end of Section 5.5.]
Owner and Design-Builder agree that the maximum aggregate liability Design der has for liquidated damages that may be assessed under this Agreement for failure to achieve the Contract (s) shapeDollars (\$).
<b>5.6 Early Completion Bonus.</b> If Substantial Completion is attained on or before
() days before the Scheduled Substantial Completion ate (the "Bonus Date"), all pay Design-Builder at the time of Final Payment under Section 7.3 hereof an completion bonus of Dollars (\$) for each day that Substantial Comple is attained earlier than the Bonus
Date. (If a GMP is not established upon execution of this Agreement, the par ronsider of the early completion bonus is applicable to any dates set for in Section 7.2 or 5.2 reof, this sum of 5.6 will need to be modified accordingly.)
[The Parties may also desire to cap the early \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \
Owner and Design-Builder agree that \ imum ag
5.7 [The Parties may also desire to modify, sicle $\epsilon$ of the General Conditions of Contract relative to compensability of delays that $\epsilon$ see the Cont. It Time, so be extended. In such case, the following option can be used.]
In addition to Design-Builder that to a cension for those events set forth in Section 8.2.1 of the General Conditions of Contract, Design-Builder that shall also be entitled to an appropriate adjustment of the Contract Price for those events. The in Section 8.2 of the General Conditions of Contract, provided, however, for Force Majeure Events, Des Builde Contract Price if said events exceed
[Check one box only]
dollars a day for each day work is delayed beyond the Scheduled
or
direct costs and expenses Design-Builder can demonstrate it has reasonably and actually incurred as a result of such event.

# Article 6 Contract Price

#### 6.0 Contract Price.

- **6.0.1** Owner shall pay Design-Builder in accordance with Article 6 of the General Conditions of Contract a contract price ("Contract Price") equal to Design-Builder's Fee (as defined in Section 6 Pereof) plus the Cost of the Work (as defined in Section 6.3 hereof), subject to any GMP established ection 6.6 hereof and any adjustments made in accordance with the General Conditions of Contract
- **6.0.2** For the specific Work set forth below, Owner agrees to pay Design-E or, as part of a Contract Price, on the following basis: (This is an optional section intended to provide the parties with ibility of unity and price limited preliminary services, such as a lump sum or cost-plus arrangement for preliminary design, program or services necess to enable Design-Builder to furnish Owner with a GMP before execution of this Agreement.)

6.1	Design-	Builder's Fee.
	6.1.1	Design-Builder's Fee shall be:
		'Choose c \ft\ \lowing:]
		accordance with Section 6.2.2 . 'ovL 's (\$), as adjusted in
		or
		pcent (
	6.1.2	De ¬-Builder's Fee win ¬djusted as rollows for any changes in the Work:
		6.1 1 For a Change coers, including additive Change Orders arising from both additive and ductive and ductive agreed that Design-Builder shall receive a Fee of
		<b>2</b> Fc eductive Change Orders, including deductive Change Orders arising from both addi and deductive items, the deductive amounts shall include:
		[Check one box only]
		No additional reduction to account for Design-Builder's Fee or any other markup.
		or

%) applied to the direct costs of the net reduction (which amount will account for a

\_ percent (\_

An amount equal to the sum of: (a) \_\_\_\_

	reduction associated with Design-Builder's Fee); plus (b) any other markups set forth in Exhibithereto applied to the direct costs of the net reduction.
	f the Work. The term Cost of the Work shall mean costs reasonably and actually incurred by Design-roper performance of the Work. The Cost of the Work shall include only the following:
Builde	Wages of direct employees of Design-Builder performing the Work at the Site or, with Owner's ment, at locations off the Site; provided, however, that the costs for those em each of Design-r performing design services shall be calculated on the basis of prevailing meat rates for design sionals performing such services or, if applicable, those rates set forth in an each of this Agreement.
•	Wages or salaries of Design-Builder's supervisory and administrative resonnel aged in the mance of the Work and who are located at the Site or working off-Site to the production cortation of material and equipment necessary for the Work.
and pe Design	Wages or salaries of Design-Builder's personnel static and at Design-Builder's price of the extent said personnel are identified in E. The erforming the function set forth in said Exhibit. The reimb to the costs of personnel stationed at an end of the extent said personnel stationed at the extent said personnel station and the ex
Design	Costs incurred by Design-Builder for envee bence, premiums, taxes, insurance, contributions seessments required by law, collictive backing prements, or which are customarily paid by a-Builder, to the extent such collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages ages and salaries paid to employees of Design-ricovered under Sections 6.3.1 to the collisions ages ages and salaries paid to employees ages ages ages ages ages ages ages a
[in	lieu of the language in Section 6.3. ່bov sign-Bu. r and Owner may want to include the
 the wa 6.3.3 h	A multiplier of
<b>6.2.5</b> persor	The ease portion of the post of travel, accommodations and meals for Design-Builder's and new sarily and the performance of the Work.
	Payme perly made by Design-Builder to Subcontractors and Design Consultants for mance of tions of the Work, including any insurance and bond premiums incurred by ntractors are resign Consultants.
	ncuri by Design-Builder in repairing or correcting defective, damaged or nonconforming (includi any warranty or corrective Work performed after Substantial Completion), provided that Vork wood on the reasonable control of Design-Builder, or caused by the ordinary mistakes or

and supplies incorporated or reasonably used in completing the Work.

inadverten and not the negligence, of Design-Builder or those working by or through Design-Builder. If

Design-Builder shall exercise its best efforts to obtain recovery from the appropriate source and provide a

sociated with such Work are recoverable from insurance, Subcontractors or Design Consultants,

Costs, including transportation, inspection, testing, storage and handling, of materials, equipment

Page 8

credit to Owner if recovery is obtained.

- **6.2.9** Costs (less salvage value) of materials, supplies, temporary facilities, machinery, equipment and hand tools not customarily owned by the workers that are not fully consumed in the performance of the Work and which remain the property of Design-Builder, including the costs of transporting, inspecting, testing, handling, installing, maintaining, dismantling and removing such items.
- **6.2.10** Costs of removal of debris and waste from the Site.
- **6.2.11** The reasonable costs and expenses incurred in establishing, operating and c' subilizing the Site office, including the cost of facsimile transmissions, long-distance telephone call stage and express delivery charges, telephone service, photocopying and reasonable petty cash express.
- **6.2.12** Rental charges and the costs of transportation, installation, minor airs and placements, dismantling and removal of temporary facilities, machinery, equipment and had not customarily owned by the workers, which are provided by Design-Builder at the Site, whether the from Design-Builder or others, and incurred in the performance of the Work.
- **6.2.13** Premiums for insurance and bonds required by this Agram on the performance of the Work.
- **6.2.14** All fuel and utility costs incurred in the perfor the Wc
- **6.2.15** Sales, use or similar taxes, tariffs or du' incurred in toperformance Work.
- **6.2.16** Legal costs, court costs and costs of i. intion a urbitration reasonably arising from Design-Builder's performance of the Work, in index such. It is not arise from disputes between Owner and Design-Builder.
- **6.2.17** Costs for permits, royalties, linea is incurred by Design-Builder as a requirement of the Contract Documents.
- **6.2.18** The cost of dencing or claim for infringement of patent rights arising from the use of a particular design, process, roductive by Owner, paying legal judgments against Design-Builder resulting from such suits or claim and pay. lements made with Owner's consent.
- **6.2.19** Do not have lost, exc. to the extent caused by Design-Builder's negligence.
- **6.2.20** Cost. curred in a damage, injury or loss in case of an emergency affecting the safety of persons and pullerty
- **6.2.21** Account. and data processing costs related to the Work.
- cost. asonably and properly incurred in the performance of the Work to the extent oved in ving Ly Owner.

# [Design-B' er and Owner may want to consider adding the following Section 6.3.23 to address the payment of warranty work:]

	<b>6.2.23</b> Owner and Design-Builder agree that	an escrow account in the amount of
	Dollars (\$	) shall be established prior to Final Completion, which
escrow	shall be used to reimburse Design-Builder for t	the Costs of the Work incurred after Final Completion
to perfo	orm warranty Work. The escrow agreement wi	Il provide that any sums not used at the expiration of
the war	ranty period shall be returned to Owner, subj	ject to any savings Design-Builder may be entitled to
under th	his Agreement. In the event the warranty escr	ow account is exhausted, but funds remain under the
GMP, O	wner shall be obligated to pay	

© 2010 Design-Build Institute of America

Design-Builder the Costs of the Work incurred after Final Completion to perform warranty Work up to the GMP.

#### 6.3 Allowance Items and Allowance Values.

- Any and all Allowance Items, as well as their corresponding Allowance Values, are set forth in the GMP Exhibit or GMP Proposal and are included within the GMP.
- 6.3.2 Design-Builder and Owner have worked together to review the Allowang .ems and Allowance Values based on design information then available to determine that the All ance Values Institute reasonable estimates for the Allowance Items. Design-Builder and Owner continue w' ig closely together during the preparation of the design to develop Construction Docu-`ts cor ent with the Allowance Values. Nothing herein is intended in any way to constitute a guarante ر gn-Builder th the Allowance Item in question can be performed for the Allowance Value.
- 6.3.3 No work shall be performed on any Allowance Ite without Design-Builder five ining in writing advanced authorization to proceed from Owner. Owner a, sthat if Design-Builder is corprovided written authorization to proceed on an Allowance Item by the date a parth in the Project achedule, due to no fault of Design-Builder, Design-Builder may be er n adju. Int of the Confact Time(s) and Contract Price.
- 6.3.4 The Allowance Value for an Allowan. om includes for direct cost of la for, materials, equipment, transportation, taxes and insurance associated v. 'he apr' design fees, Design-Builder's overall r and general conditions costs, overhead and fee, ct mana<sub>b</sub> are deemed to be included in the original ntract Pi and are not subject to adjustment, regardless of the actual amount of the Allowance Ite.

	In the alternative, the parties ma	vw. 'toa.	Section 6.4.4 and add the	following provision.
--	------------------------------------	-----------	---------------------------	----------------------

	In the event ι	76. dir	rect cos	f labor,	aterials, e	quipment,	transportatio	n, taxes	and
insuran	ce associated with	د اا۱۲۰۰	∵∘m i.				perc	ent (	
%) grea	ter than or less tha	n tı. ''ov	vance	วr such	Allowance	Item, Desig	n-Builder and	Owner a	gree
that De	sign-Quilder's right	to Fee	¹ markup s	be ad آنہ ر	iusted in ac	cordance w	ith Section 6.	2.2.	

inal costs fo. Allowance Item is more than or less than the stated Allowance Value, the C ract Price adjusted accordingly by Change Order, subject to Section 6.4.4. The amount of the Nange der sha. Alect the difference between actual costs incurred by Design-Builder the particula ' Jance Item and the Allowance Value.

#### Non-Reimbursable sts.

- The owing shall not be deemed as costs of the Work:
- Compensation for Design-Builder's personnel stationed at Design-Builder's principal or .ch offices, except as provided for in Sections 6.3.1, 6.3.2 and 6.3.3 hereof.
- **6.4.1.2** Overhead and general expenses, except as provided for in Section 6.3 hereof, or which may be recoverable for changes to the Work.
- **6.4.1.3** The cost of Design-Builder's capital used in the performance of the Work.
- **6.4.1.4** If the parties have agreed on a GMP, costs that would cause the GMP, as adjusted in accordance with the Contract Documents, to be exceeded.

[The parties shall comply with the following Section 6.6 based upon whether the GMP is agreed upon before the execution of this Agreement or will be developed and agreed upon after execution of this Agreement. If the parties do not use a GMP, this Section 6.6 shall be deemed inapplicable and compensation to Design-Builder shall be based on those fees and costs identified in the balance of this Article 6.]

6.5 The Guaranteed Maximum Price ("GMP").

6.5.1

6.5.1.1 Design-Builder guarantees that it shall not exceed the GM.  Dollars (\$	
Design-Builder guarantees to shall not sed the GMP of	ant specific line it ent de overruns de lder agre owe , tha which exceet sMP, a which will be develoked in advance
	// ///////////////////////////////////
specific line item provided as the GMP, wided, however, that it do item for its general project man tender and general conditions costs, in the Dollars (\$	sis for the GMP shall be er does not guarantee and it does guarantee the line in the amount of
The GMP include. Contingency in the amount of	usive use for unanticipated the Contract Documents. But calation of materials; (does not on 8.2.2 of the General to but do not result in all

but I limited to changes in scope or any other item which would enable Design-Builder to increate the GMP under the Contract Documents. Design-Builder shall provide Owner notice of all are pated charges against the Contingency, and shall provide Owner as part of the monthly status port required by Section 2.1.2 of the General Conditions of Contract an accounting of the Contingency, including all reasonably foreseen uses or potential uses of the Contingency in the upcoming three (3) months. Design-Builder agrees that with respect to any expenditure from the Contingency relating to a Subcontractor default or an event for which insurance or bond may provide reimbursement, Design-Builder will in good faith exercise reasonable steps to obtain performance from the Subcontractor and/or recovery from any surety or insurance company.

Design-Builder agrees that if Design-Builder is

subsequently reimbursed for said costs, then said recovery will be credited back to the Contingency.

- 6.5.2 GMP Established after Execution of this Agreement.
  - 6.5.2.1 GMP Proposal. If requested by Owner, Design-Builder shall submit a GMP Proposal to Owner which shall include the following, unless the parties mutually agree otherwise:
    - 6.5.2.1.1 A proposed GMP, which shall be the sum of:
      - i. Design-Builder's Fee as defined in Section 6.2
      - ii. The estimated Cost of the Work as defined in Sc reof, inclusiv of any Design-Builder's Contingency as defined in n 6.6.1.2 her and
      - iii. If applicable, any prices estal and under Section 6.1.2 hear
    - 6.5.2.1.2 ray include, by ay of example, The Basis of Design Do ts, which Owner's Project Criteria, which thed to the GMP nd are 1 set ic in det Proposal;
    - 6.5.2.1.3 ntions and arifications made by Design-Builder in the A list of the as. st is intended to supplement the information preparation of the GMP Propos vhir' contained in the d. cions and is specifically included as part of the s and sp. Basis of Design Docu. 1.
    - 6.5.2.1.4 The Sched d Su intial Co. Letion Date upon which the proposed GMP is base `n the extent so ' date not already been established under Section 5.2.1 bedule upc which it is cheduled Substantial Completion Date is based; herec
    - 6.5.2.1.5 oplica. of Allowance Items, Allowance Values, and a statement of their basis;
    - , a schedule of alternate prices; If applic
    - 6.5 It applicable, a schedule of unit prices;
    - **5.2.1.8** If applicable, a statement of Additional Services which may be performed which are not included in the GMP and which, if performed, shall be the basis for an ase in the GMP and/or Contract Time(s); and
    - 6.5.2.1.9 The time limit for acceptance of the GMP Proposal.
  - **2.2** Review and Adjustment to GMP Proposal. After submission of the GMP Proposal, besign-Builder and Owner shall meet to discuss and review the GMP Proposal. If Owner has any comments regarding the GMP Proposal, or finds any inconsistencies or inaccuracies in the information presented, it shall promptly give written notice to Design-Builder of such comments or findings. If appropriate, Design-Builder shall, upon receipt of Owner's notice, make appropriate adjustments to the GMP Proposal.

- **6.5.2.3** Acceptance of GMP Proposal. If Owner accepts the GMP Proposal, as may be amended by Design-Builder, the GMP and its basis shall be set forth in an amendment to this Agreement.
- **6.5.2.4** Failure to Accept the GMP Proposal. If Owner rejects the GMP Proposal, or fails to notify Design-Builder in writing on or before the date specified in the GMP Proposal that it accepts the GMP Proposal, the GMP Proposal shall be deemed withdrawn and of no effect. In such event, Owner and Design-Builder shall meet and confer as to how the Project will proceed, with Owner having the following options:
  - **6.5.2.4.1** Owner may suggest modifications to the GMP Pro al, whereup n, if such modifications are accepted in writing by Design-Builder, GMP Prop shall be deemed accepted and the parties shall proceed in accordance h Section 5.2.3 above;
  - **6.5.2.4.2** Owner may authorize Design-Builder to continue to p ed with the V on the basis of reimbursement as provided in Section 6.1 hereof with GMP, i wich case all references in this Agreement to the C P shall not be applicable,
  - **6.5.2.4.3** Owner may terminate this Agreem for convenience in accordance with Article 8 hereof; provided, however frent, Light and are also accordance with the payment provided for in Section 3.2 here

If Owner fails to exercise any of the ve options, D n-Builder shall ave the right to (i) continue with the Work as if Own, and electron proceed in accordance with Item 6.6.2.4.2 above, and be paid hower action unless and until Owner notifies it in writing to stop the Work, or (ii) suspen armance with in accordance with Section 11.3.1 of the General Conditions. Contract, evided, however, that in such event Design-Builder shall not be entitled to the paint provide for in Section 8.2 hereof.

#### 6.5.3 Savings.

**6.5.3.1** If the sun. The acceptable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, any prices established unconstant the Work and Design-Builder's Fee (and, if applicable, and if applicable, and if applicable, and if applicable unconstant the Work and Design-Builder's Fee (and, if applicable, and if applicable, and if applicable, and if applicable unconstant the Work and Design-Builder's Fee (and, if applicable, and i

noose one of the following:]

	percent	(	%) to Design-Bui	lder
and _	percent (	%) to Owner.		
		or		
	Th .rst	Dollars (\$	)of	f Savings
shall	provided to (choose either Design-Buil	der or Owner)		
with	e balance of Savings, if any, shared		percent (	%)
to	Jign-Builder and	percent (	%) to O	wner.
	.2 Savings shall be calculated and paid a	•		

would have been payable to Design-Builder as a Cost of the Work, the parties shall recalculate the Savings in light of the costs so incurred, and Design-Builder shall be paid by Owner accordingly.

DBIA Document No. 530 Page **13** 

#### 6.6 **Performance Incentives**

Owner and Design-Builder have agreed to the performance incentive arrangements set forth in Exhibit .

[The parties are encouraged to discuss and agree upon performance incentives that will influence project success. These incentives may consist of Award Fees, incentives for safety, personnel retention, client satisfaction and similar items.]

# **Article 7**

	Procedure for Payment
7.1	Progress Payments.
	7.1.1 Design-Builder shall submit to Owner on the
	<b>7.1.2</b> Owner shall make payment within ten ', days after ner's rece of an properly submitted and accurate Application for Payment in accounce with Arti 6 of the Gen a Conditions of Contract, but in each case less the total of payments properly withheld under Section 6.3 of the General Conditions of Contract.
	<b>7.1.3</b> If Design-Builder's Fee under S ic 2.1 herec is fixed amount, the amount of Design-Builder's Fee to be included in Design-Builder's on. Applicate for Payment and paid by Owner shall be proportional to the percentage of the Wick co. In ted, less payments previously made on account of Design-Builder's Fee.
7.2	Retainage on Progress Pay. *s.
	Payment powever, that work fifty percent (50%) of the Work has been satisfactorily completed by Design-Boulet and Additional Mounts from Design-Builder's subsequent Applications for Payment.  Owner will also has any consider reducing retainage for Subcontractors completing their work early in Project.
	[Design-Builder d Owner may want to consider substituting the following retainage provision.]
	Own will recain percent (
	<b>7.2.2</b> Within fifteen (15) days after Substantial Completion of the entire Work or, if applicable, any portion of the Work, pursuant to Section 6.6 of the General Conditions of Contract, Owner shall release to

noted in the Certificate of Substantial Completion; and (b) all

Design-Builder all retained amounts relating, as applicable, to the entire Work or completed portion of the Work, less an amount equal to: (a) the reasonable value of all remaining or incomplete items of Work as

other amounts Owner is entitled to withhold pursuant to Section 6.3 of the General Conditions of Contract.

#### [If Owner and Design-Builder have established a warranty reserve pursuant to Section 6.3.23 above, the following provision should be included.]

If a warranty reserve has been established pursuant to Section 6.3.23 above, Owner shall at the time of Substantial Completion retain the agreed-upon amounts and establish an escrow account as contemplated by Section 6.3.24 above.

7.3 **Final Payment.** Design-Builder shall submit its Final Application for Payment to C ar in accordance with Section 6.7 of the General Conditions of Contract. Owner shall make payment or -sign-Builde roperly et aside for submitted and accurate Final Application for Payment (less any amount the parties ma, 'e agreed warranty work) within ten (10) days after Owner's receipt of the Final Application for Payn. pro d that Design Builder has satisfied the requirements for final payment set forth in Section 6.7.2 of the ral Condition Contract.

<b>7.4</b> Interest. Payments due and unpaid by Owner to Design-Bu.	whether pro	ogress payı.	or final
payment, shall bear interest commencing five (5) days after payment is duc	he rate of		
percent (%) per month un+''			

7.5 Record Keeping and Finance Controls. Design .ider ack vledges t thir greement is to be administered on an "open book" arrangement relative Costs of the Vork. Design ∡er shall keep full and detailed accounts and exercise such controls as may be 'essary for pr r financial man, gement, using accounting and control systems in accordance with generally acce, accour g principles and as may be provided in the Contract Documents. During the performance ካe Work f \_\_\_, period of three (3) years after Final Payment, Owner and Owner's accountants shall be at. nd the right to audit from time-to-time, upon access i reasonable notice, Design-Builder's records, L. N. rrespon. e, receipts, subcontracts, purchase orders, vouchers, memoranda and other data relating to Woodl of who esign-Builder shall preserve for a period of three (3) years after Final Payment. Such inspec in she like place at Design-Builder's offices during normal business hours unless another Ir nd time is age ed to by a parties. Any multipliers or markups agreed to by the Owner and Design-Builder as `greeme are only subject to audit to confirm that such multiplier or 01 . markup has been charged in accord. witn. are nent, with the composition of such multiplier or markup not being subject to audit.

## Article 8

#### Termination for Convenience

- Upon ten (10) da written notice to Design-Builder, Owner may, for its convenience and without cause, ect to terminate this Agre ent. In such event, Owner shall pay Design-Builder for the following:
  - All W cexecuted and for proven loss, cost or expense in connection with the Work;
  - 8.1.2 asonable costs and expenses attributable to such termination, including demobilization cts and .ounts due in settlement of terminated contracts with Subcontractors and Design Consultants;
  - 8.1.3 [Choose one of the following:]

Г	٦	The f	air and	reasonab	e sums	for	overhead	and	profit	on	the	sum	of it	ems	8.1.1	and
8.	1	.2 abo	ve.													

	or		
	Overhead and profit in the amount of on the sum of items 8.1.1 and 8.1.2 above.	percent (	%)
<b>8.2</b> the follo	In addition to the amounts set forth in Section 8.1 above, Design-Bowing as applicable:	uilder shall be entitled to rece	ive one of
	<b>8.2.1</b> If Owner terminates this Agreement prior to commencembe paid _ percent (	maining balar of the Contr	ct Price,
	8.2.2 If Owner terminates this Agreement after commencement paid percent (%) of the remaining behavior, that if a GMP has not been established, the above p balance of the most recent estimated Contract Price.	valance of the Contra rice,	pr ued,
forth in S		e ai foceeds to dog and to use Work fouct shall others wi sign-Builder	be as set
_		er agree to establish their resp ing the performance of the Pr	
9.1	Owner's Resentatives.		
	9.1.1 Ow design the individual listed below as its strength of the authority and responsible under Section 2.3 and General Conditions of Contract: (Identification)	pility for avoiding and resolving	g disputes
	<b>9.1.2</b> Own designates the individual listed below as its Owne the author and responsibility set forth in Section 3.4 of the consideration of the considerat		

9.2	Design-Builder	's Rep	oresentatives

- 9.2.1 Design-Builder designates the individual listed below as its Senior Representative ("Design-Builder's Senior Representative"), which individual has the authority and responsibility for avoiding and resolving disputes under Section 10.2.3 of the General Conditions of Contract: (Identify individual's name, title, address and telephone numbers)
- 9.2.2 Design-Builder designates the individual listed below as its Design-Builder Repr/ tative, which individual has the authority and responsibility set forth in Section 2.1.1 of the al Conditions **Contract:** (Identify individual's name, title, address and telephone numbers)

#### and Ins ice

- 10.1 Insurance. Design-Builder and Owner s II p re the . ance coverages set forth in the Insurance Exhibit attached hereto and in accordance with the Seneral Conditions of Contract.
- 10.2 Bonds and Other Per. esign-Builder shall provide the following ^∘curity. performance bond and labor and me 'al pay ' hol or other performance security:

Performance Cond.
[Check one wonly. 13 is checkewihen no bond is required.]
R ired Not Required
r 👡 lent Bond.
[Cher' `x onl, `no box is checked, then no bond is required.]
☐ H uired ☐ Not Required
Other Performance Security.
eck one box only. If no box is checked, then no other performance security is required. If the "Required" box is checked, identify below the specific performance security that is being required and all salient commercial terms associated with that security.]
Required Not Required

# Article 11 Other Provisions

11.1 Other provisions, if any, are as follows: (Insert any additional provisions)

[Section 2.3.1 of the General Conditions of Contract sets forth a tradition orgligence indard as i relates to the Design-Builder's performance of design professional service. the isis of Design
Documents identify specific performance standards that can be objectively me, the parties including the following language, agree that the Design-Builder is obligated to achiev the stand]
Notwithstanding Section 2.3.1 of the General Condition. Contract, if the parties agree upon specific performance standards in the Basis of Design Parts, the sign profession ervices shall be performed to achieve such standards.
[In lieu of Sections 10.3.1 through 10.3.3 of the parties may want to delete such sections and inc' le the for permitting dispute resolution clause.]
Any claims, disputes, or consider sets between the parties arising out of or related to the Agreement, or the breach thereof, which has not been lived in accordance with the procedures set for the General Constitution. So Contractional Be resolved in a court of competen jurisdiction in the state in which the Project local

## REQUEST FOR QUALIFICATIONS (RFQ)

In executing this Agreement, Owner and Design-Builder each individually represents that it has the necessary financial resources to fulfill its obligations under this Agreement, and each has the necessary corporate approvals to execute this Agreement, and perform the services described herein.

OWNER:	DESIGN-BUILDER:
(Name of Owner)	(Name of Design-Builder)
(Signature)	(Signature)
(Printed Name)	(Printed Name)
(Title)	(T'''
Date:	Date:

Caution: An original DBIA document has this cauting and in bluctiss is a printable copy and an original assures that changes will not be obscured as may occur with a new produced.



# STANDARD FORM L GF'LERAL CONDITIONS OF CONTRACT L GWEL OWNER AND DESIGN-BUILDER



# Design-Build Institute of America - Contract Documents LICENSE AGREEMENT

By using the DBIA Contract Documents, you agree to and are bound by the terms of this License Agreement.

- 1. License. The Design-Build Institute of America ("DBIA") provides DBIA Contract Documents and Lenses their use worldwide. You acknowledge that DBIA Contract Documents are protected by the copyright laws of the Liced States. You are a limited nonexclusive license to: (a) Use DBIA Contract Documents on any number of machines Led, lease License to rented by your company or organization; (b) Use DBIA Contract Documents in printed form for bona fide Linear Losses; and (c) Topy DBIA Contract Documents into any machine-readable or printed form for backup or modification. Losses in support Livense your permitted use.
- 2. User Responsibility. You assume sole responsibility for the selection of specific documents or portions to chieve your intended results, and for the installation, use, and results obtained from a BIA Contract Documents ⊿ acknowledge that you understand that the text of the DBIA Contract Documents has imporlegal consequences and that consultation You will not reasont that any of the with an attorney is recommended with respect to use or modificf the t⊾ contract documents you generate from DBIA Contract Documer are DL nless (ع ع document text is used 'ocume. without alteration or (b) all additions and changes to, and de' ons from, t :ext are c 'v' vn.
- 3. Copies. You may not use, copy, modify, or transfer DBIA cract Docume , or any copy, pdification or merged portion, in whole or in part, except as expressly provided for in this anse. Repression of DBIA Contract Documents in printed or machine-readable format for resale or educational aurposes are prohibited. You will reproduce and include DBIA's copyright notice on any printed or machine-reaction, or portion merged into another document or program.
- 4. Transfers. You may not transfer possession of any oy, dification merged portion of DBIA Contract Documents to another party, except that a party with whom you are onto any may revive and use such transferred material solely for purposes of its contract with you. You wonot sublice assign transfer this license except as expressly provided in this Agreement, and any attempt to do
- **5. Term.** The license is effective for one year from a line of license. DBIA may elect to terminate it earlier, by written notice to you, if you fail to comply with any term. Condition of Agreement.
- 6. Limited Warranty. DBIA warrants the electron files or other media by which DBIA Contract Documents are furnished to be free from defects in more and workmanship der normal use during the Term. There is no other warranty of any kind, expressed or implied, it iding, and limited to elimplied warranties of merchantability and fitness for a particular purpose. Some states do to allow to the ion of implied warranties, so the above exclusion may not apply to you. This warranty gives you specific gal rices and youngalso have other rights which vary from state to state. DBIA does not warrant the other particular purpose. Some states do to allow the contract Documents will be affected by the property of the property
- 7. Lim lions of Remedies. DBL entire liability and your exclusive remedy shall be: the replacement of any document not more ing DBIA's "' 'arrai 'which is returned to DBIA with a copy of your receipt, or at DBIA's election, your money will a refure in note into which is returned to DBIA with a copy of your receipt, or at DBIA's election, your money will a refure in note into which is returned to DBIA with a copy of your receipt, or at DBIA's election, your money will a refure in note in the will be liable to you for any damages, including any lost profits, lost savings or other inclusions arising out of the use or inability to use DBIA Contract Documents even if DBIA has been advised of the possibility such damages, or for any claim by any other party. Some states do not allow the limitation or exclusion of liability for cidental or consequential damages, so the above limitation or exclusion may not apply to you.
- **8.** Acknowledge that you have read this agreement, understand it and agree to be bound by its terms and conditions and that it will be governed by the laws of the District of Columbia. You further agree that it is the complete and exclusive statement of your agreement with DBIA which supersedes any proposal or prior agreement, oral or written, and any other communications between the parties relating to the subject matter of this agreement.

## REQUEST FOR QUALIFICATIONS (RFQ)

## **INSTRUCTIONS**

For DBIA Document No. 535 Standard Form of General Conditions of Contract Between Owner and Design-Builder (2010 Edition)

#### **General Instructions**

No.	Subject	Instruction
1.	Standard Forms	Standard form contracts have long served an important function in the United States and international construction markets. The common purpose of these for to provide an economical and convenient way for parties to contract for design and construction in services. As standard forms gain acceptance and are used with increased frequency, parties able to enter contracts with greater certainty as to their rights and responsibilities.
2.	DBIA Standard Form Contract Documents	Since its formation in 1993, the Design-Build Institute of America (Db. regularly eval ad the needs of owners, design-builders, and other parties to the design-build cess in proation for developing its own contract forms. Consistent with DBIA's mission of promuting housign-build practices, DBIA believes that the design-build count should reflect a balance. Each to risk that considers the legitimate interests of all parties to design-build process. DBIA's Standard Form Contract Documents reflect a modern risk allocation and hallocating end housign-build practices.
3.	Use of Non- DBIA Documents	To avoid inconsistencies amor cuments used the same prop., DBIA's Standard Form Contract Documents should not be us conjund with non-DBIA documents unless the non-DBIA documents are approprintely mou on advice of legal counsel. Moreover, care should also be taken when using different editions and DBIA Standard Form Document on the same project to ensure consistency.
4.	Legal Consequences	DBIA Standard Form Corract cuments legally binding contracts with important legal consequences. Contracting intries individual dencouraged to seek legal counsel in completing or modification.
5.	Reproduction	DBIA hereby nts to process a limited license to reproduce its Documents consistent with the License Agreement accompany, these Documents. At least two original versions of the Agreement should be signed to a parties. Any other reproduction of DBIA Documents is strictly prohibited.
	Modif	Effective a is accomplished when the parties give specific thought to their contracting goals and the cailor the intract to meet the unique needs of the project and the design-build team. For ason, these Documents may require modification for various purposes including, for example, omply with local codes and laws, or to add special terms. DBIA's latest revisions to its Documents poide the parties an opportunity to customize their contractual relationship by selecting various operated and contract clauses that may better reflect the unique needs and risks associated with the project.  Any modifications to these Documents should be initialed by the parties. At no time should a document be re-typed in its entirety. Re-creating the document violates copyright laws and destroys one of the advantages of standard forms-familiarity with the terms.
7.	Execution	It is good practice to execute two original copies of the Agreement. Only persons authorized to sign for the contracting parties may execute the Agreement.

## **Specific Instructions**

Section	Title	Instruction
General	Purpose of This Document	The General Conditions of Contract provide the terms and conditions under which the Work of the Project will be performed.  This document accompanies DBIA Document No. 525 and DBIA Document No. 530 (each referred to herein generally as "Agreement"). It may also be incorporated by reference into other related agreements, as between the Design-Builder and the Design Consultant, ar "the Design-Builder and the Subcontractor.
General	Checklist	The following Sections reference documents that are to be attac' to the Agreement:  Section 3.5.1 Owner's Permit List Article 5 Insurance and Bonds Section 9.4.2 Unit Prices
2.1.3	Schedule	The parties are encouraged, if possible, to ag to a schedule for the exection of the Agreement or upon establish. 'he GMP.
2.2.1	Design Professional Services	The parties should be aware that in to requ. compliance v state licensing laws for design professionals, some state licensing laws for that to resign professional have a corporate professional license.
2.3.1	Standard of Care for Design Professional's Services	Design-Builder's obligation of deliver design that meets prevailing industry standards. However, DBIA has provided design that meets prevailing industry standards. However, DBIA has provided design that can be objectively measured, Design-Builder is objectively measured, Design-Builder
3.5.1	Governme Approvals 1 Permits	Design-Bul. is respond obtaining all necessary permits, approvals and licenses, except to the extent spond permits, approvals, and licenses are set forth in an Owner's Permit List, which rist be attached in exhibit to the Agreement. The parties, prior to execution of the Agreement, iscuss which mits, approvals and licenses need to be obtained for the Project and which party in the position to do so.
5.1.1	Design-Builder's rurance Requirements	D sn-Builder is obligated to provide insurance coverage from insurance carriers that meet the ceria set forth in the Insurance Exhibit attached to Section 10.1 of the Agreement.
5.1	Jesign-Build	ties are advised that their standard insurance policies may contain exclusions for the design- build delivery method. This Section 5.1.2 requires that any such exclusions be deleted from the policy.
5.2	Owner' Requirements	Owner, in addition to providing the insurance set forth in this Section and Section 5.3, is also obligated to procure the insurance coverages for the amounts and consistent with the terms set forth in the Insurance Exhibit made part of the Agreement.
5.4	Bonds and Other Performance Security	Design-Builder is only obligated to provide bonds or other forms of performance security to the extent called for in Section 10.2 of the Agreement.

## REQUEST FOR QUALIFICATIONS (RFQ)

Section	Title	Instruction				
8.2.2	Compensability for Force Majeure Events	The parties are provided the option in the Agreement of negotiating whether the Design- Builder is entitled to compensation for Force Majeure Events.				
9.4.1	Contract Price Adjustments	Unit prices, if established, shall be attached pursuant to Article 2 of the Agreement.				
9.4.3	Payment/ Performance of Disputed Services	When Owner disputes Design-Builder's entitlement to a change order or disagrees with Design-Builder regarding the scope of Work, and nevertheless expects Design-Builder to perform the services, Design-Builder's cash flow and ability to complete the Word will be hampered if Owner fails to pay Design-Builder for the disputed services. This Section vovides a balanced approach whereby Design-Builder is required to perform the services. Owner is roughled to pay fifty percent (50%) of Design-Builder's reasonable estimated directors and perform the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled. By so doing, Owner does not forfeit in the dispute is settled.				
Article 10	Contract Adjustments and Disputes	DBIA endorses the use of partnering, negotia mediation and arbitration, the prevention and resolution of disputes. The General Concers of Contract provides for the parties' Representatives and Senior Represer to attend a negotiate the partie of disagreement. If this attempt fails, the dispute of dispute of dispute that cannot be resolve mediation. If then be not obtaining arbitration, unless the parties elect in the Agrament to submit or dispute to obtaining arbitration.				
10.3.4	Arbitration	The prevailing par' any arbi. all receive reasonable attorneys' fees from the other party. DBIA suppor 'oser pay. vision to encourage parties to negotiate or mediate their differences and to m. m. 'he numb. frivolous disputes.				
10.4	Duty to Continue Performance	Pending the resolution. f any nute or coagreement, both Owner and Design-Builder shall continue perform the respect duties under the Contract Documents, unless the parties provenue in the Contract Documents.				
10.5	Consequenti Damage.	DBIA belie that it is a priate for either Owner or Design-Builder to be responsible to the other for convential as ages arising from the Project. This limitation on consequential damages in no vestricts, however, the payment of liquidated damages, if any, under Article 5 Agreement.				
11.4	Design-Builder's J Terminate for Cause	If De reperty terminates the Agreement for cause, it shall recover from Owner in the server way as it owner had terminated the Agreement for convenience under Article 8 of the greement. Owner shall pay to Design-Builder its costs, reasonable overhead and profit on the osts, and an additional payment based on a percentage of the remaining balance of the Contract ice, all as more fully set forth in Article 8 of the Agreement.				
Article 12	Flectronic	Design-Builder and Owner shall agree on the software and format for the transmission of Electronic Data. Ownership of Work Product in electronic form is governed by Article 4 of the Agreement. The transmitting party disclaims all warranties with respect to the media transmitting the Electronic Data, but nothing in this Article is intended to negate duties with respect to the standard of care in creating the Electronic Data.				

## TABLE OF CONTENTS

Article	Name	Page
Article 1	General	1
Article 2	Design-Builder's Services and Responsibilities	2
Article 3	Owner's Services and Responsibilities	6
Article 4	Hazardous Conditions and Differing Site Conditions	8
Article 5	Insurance and Bonds	9
Article 6	Payment	11
Article 7	Indemnification	
Article 8	Time	د ہـ
Article 9	Changes to the Contract Price and Time	15
Article 10	Contract Adjustments and Disputes	17
Article 11	Stop Work and Termination for Cause	19
Article 12	Electronic Data	21
Article 13	Miscellaneous	22

# Article 1 General

#### 1.1 Mutual Obligations

**1.1.1** Owner and Design-Builder commit at all times to cooperate fully with each oth and proceed on the basis of trust and good faith, to permit each party to realize the benefits afford ander the Contract Documents.

#### 1.2 Basic Definitions

- **1.2.1** Agreement refers to the executed contract between Owner and Design-Bu. Inder either P Document No. 525, Standard Form of Agreement Between Owner and Design-Builde. Imp Sur ID Edition) or DBIA Document No. 530, Standard Form of Agreemer \* Between Owner and Des. 'qui' Cost Plus Fee with an Option for a Guaranteed Maximum Price (201. 'tion).
- **1.2.2** Basis of Design Documents are as follows: RIA Doc ant No. 530, and and Form of Agreement Between Owner and Design-Builder Co as re "than and for a C anteed Maximum Price, the Basis of Design Documents are those cuments so ifically lie in applicable, the GMP Exhibit or GMP Proposal as being the "Basis of a sign Docume". For DBIA and nent No. 525, Standard Form of Agreement Between Owner and Design ilder Lur aum, the Basis of Design Documents are the Owner's Project Criteria, Design-Builder's Proposal of the eviation List, if any.
- 1.2.3 Construction Documents are the numents, issisting of Drawings and Specifications, to be prepared or assembled by the Design-iloumnisten. In the Basis of Design Documents unless a deviation from the Basis of Design Documents is cifically forth in a Change Order executed by both the Owner and Design-Perior, as part of the design if we process contemplated by Section 2.4 of these General Conditions of Control of
- **1.2.4** Day or Days shall in cale. In sunless otherwise specifically noted in the Contract Documents.
- Design reant is a qualified, licensed design professional who is not an employee of Design-Builder, but is read by Design-Builder, or employed or retained by anyone under contract with Design-Builder, to furnish sign services required under the Contract Documents. A Design Sub-Consultant is a qualification sed contract of the Design Consultant, but is retained by anyone under contract to Design Consultant, to furnish sesign service equired under the Contract Documents.
- **1.2.7** *F' Completion* is the date on which all Work is complete in accordance with the Contract ..., including but not limited to, any items identified in the punch list prepared under Section 6.6.1 and the submission of all documents set forth in Section 6.7.2.
- **1.2.8** Force Majeure Events are those events that are beyond the control of both Design-Builder and Owner, including the events of war, floods, labor disputes, earthquakes, epidemics, adverse weather conditions not reasonably anticipated, and other acts of God.
- **1.2.9** *General Conditions of Contract* refer to this DBIA Document No. 535, *Standard Form of General Conditions of Contract Between Owner and Design-Builder* (2010 Edition).
- 1.2.10 GMP Exhibit means that exhibit attached to DBIA Document No. 530, Standard Form of

Agreement Between Owner and Design-Builder - Cost Plus Fee With an Option for a Guaranteed Maximum Price, which exhibit will have been agreed upon by Owner and Design-Builder prior to the execution of the Agreement.

- **1.2.11** *GMP Proposal* means that proposal developed by Design-Builder in accordance with Section 6.6 of DBIA Document No. 530, Standard Form of Agreement Between Owner and Design-Builder Cost Plus Fee With an Option for a Guaranteed Maximum Price.
- **1.2.12** Hazardous Conditions are any materials, wastes, substances and chermals deemed to be hazardous under applicable Legal Requirements, or the handling, storage, remedian, or disposal which are regulated by applicable Legal Requirements.
- **1.2.13** Legal Requirements are all applicable federal, state and local laws, containances, rule regulations, orders and decrees of any government or quasi-government entity having isdiction over a Project or Site, the practices involved in the Project or Site, or any Work.
- **1.2.14** Owner's Project Criteria are developed by or for Owner. scribe Owner's program i ,uirements expandability requirements, as well and objectives for the Project, including use, space, price, time, site as submittal requirements and other requirements g der's perfor າesign ∙ ce of the Work. sign performance Owner's Project Criteria may include concept docum , desig. riteria specifications, design specifications, and LEEP of other sus ia and other Projectable desi specific technical materials and requirement.
- **1.2.15** Site is the land or premises or ich the F or Jocated.
- **1.2.16** Subcontractor is any person of the Work and shall the sterialment of suppliers.
- **1.2.17** Sub-Subcontro any person rentil, relationed by a Subcontractor as an independent contractor to perform any person rector's cork and shall include materialmen and suppliers.
- **1.2.18** Substantial Completic. Substant. Implete means the date on which the Work, or an agreed upon porticity of the Work, is suit in the contract Documents so that Owner can cup it is the Project a portion thereof for its intended purposes.
- **1.2.19** Wor, comprire comprire comprises and labor contract Documents. Cluding procuring and furnishing all materials, equipment, services and labor composition on the Contract Documents.

#### **Article 2**

#### **Design-Builder's Services and Responsibilities**

#### 2.1 Coneral ces.

- **2.1.1** Design-Builder's Representative shall be reasonably available to Owner and shall have the necessary expertise and experience required to supervise the Work. Design-Builder's Representative shall communicate regularly with Owner and shall be vested with the authority to act on behalf of Design-Builder. Design-Builder's Representative may be replaced only with the mutual agreement of Owner and Design-Builder.
- 2.1.2 Design-Builder shall provide Owner with a monthly status report detailing the progress of

the Work, including (i) whether the Work is proceeding according to schedule, (ii) whether discrepancies, conflicts, or ambiguities exist in the Contract Documents that require resolution,

- (iii) whether health and safety issues exist in connection with the Work; (iv) status of the contingency account to the extent provided for in the Standard Form of Agreement Between Owner and Design-Builder Cost Plus Fee with an Option for a Guaranteed Maximum Price; and
- (v) other items that require resolution so as not to jeopardize Design-Builder's ability to complete the Work for the Contract Price and within the Contract Time(s).
- Unless a schedule for the execution of the Work has been attached to the A ement as an exhibit at the time the Agreement is executed, Design-Builder shall prepare and submit, ast three (3) 'ays prior to the meeting contemplated by Section 2.1.4 hereof, a schedule for the execu . Owner's of the Wor' review and response. The schedule shall indicate the dates for the start and con. on of th ∡rious stages of Work, including the dates when Owner information and approvals are required Design-Build to achieve the Contract Time(s). The schedule shall be revised as required by condition nd progress of Work, but such revisions shall not relieve Design-Builder of its obligations to complete to 'ork wi' Contract Time(s), as such dates may be adjusted in accordar with the Contract Docu. wner's review of, and response to, the schedule shall not be construed Nieving Design-Builder of complete and exclusive control over the means, methods, sequences and tec. ues for executing the Work.
- **2.1.4** The parties will meet within seven (7) drafter extion of Agree at to discuss issues affecting the administration of the Work and amplement necessary aures, including those relating to submittals and payment, to facility ne ability of the contract Documents.

#### 2.2 Design Professional Services.

licensing laws, provide through qualified. 2.2.1 Design-Builder shall, consistent "ith licable s. Builder, or procured from qualified, independent licensed design professionals employed / De. licensed Design Consul\* he necessary sign somes, including architectural, engineering and other design professional serv. preparation of the equired drawings, specifications and other design Tu. submittals to permit Design `uilde<sub>1</sub> om, te the Work consistent with the Contract Documents. Nothing in the Contract Docu-'s is inten deemed to create any legal or contractual relationship between O' and any Design

#### 2.3 Standard of re for profession, ervices.

2.3.1 The state of the design professional services performed to execute the Work shall be are and ski charily used by members of the design profession practicing under similar conditions at the same time same time.

#### 4 Dec mer rvices.

**2.4.1** Design Builder and Owner shall, consistent with any applicable provision of the Contract Documents are upon any interim design submissions that Owner may wish to review, which interim resign submissions may include design criteria, drawings, diagrams and specifications setting forth the requirements. Interim design submissions shall be consistent with the Basis of Design Documents, as the Basis of Design Documents may have been changed through the design process set forth in this Section 2.4.1. On or about the time of the scheduled submissions, Design-Builder and Owner shall meet and confer about the submissions, with Design-Builder identifying during such meetings, among other things, the evolution of the design and any changes to the Basis of Design Documents, or, if applicable, previously submitted design submissions. Changes to the Basis of Design Documents, including those that are deemed minor changes under Section 9.3.1, shall be processed in accordance with Article 9. Minutes of the meetings, including a full listing of all changes, will be maintained by Design-Builder and provided

to all attendees for review. Following the design review meeting, Owner shall review and approve the interim design submissions and meeting minutes in a time that is consistent with the turnaround times set forth in Design-Builder's schedule.

- 2.4.2 Design-Builder shall submit to Owner Construction Documents setting forth in detail drawings and specifications describing the requirements for construction of the Work. The Construction Documents shall be consistent with the latest set of interim design submissions, as such submissions may have been modified in a design review meeting and recorded in the meetings minutes. The parties shall had a design review meeting to discuss, and Owner shall review and approve, the Construction Document and accordance with the procedures set forth in Section 2.4.1 above. Design-Builder shall procedure with construction in accordance with the approved Construction Documents and shall submit one of approved astruction Documents to Owner prior to commencement of construction.
- **2.4.3** Owner's review and approval of interim design submissions, meeting inutes, and the Construction Documents is for the purpose of mutually establishing a conformed set of Construction Documents of the Work. Neither Owner and eview nor approval of any the design submissions, meeting minutes, and Construction Documents shall be deemed to transfer any owner and liability from Design-Builder to Owner.
- **2.4.4** To the extent not prohibited by the Contactor Documents or Legarguire and Construction Documents of the Work to permit construction to proceed on that portion of Construction Documents for the entire Work.

#### 2.5 Legal Requirements.

- **2.5.1** Design-Builder shall perform the 'ork scordanc 'th all Legal Requirements and shall provide all notices applicable to the Work as requirements.
- 2.5.2 antract me(s) shoul be adjusted to compensate Design- Builder for The Contract Pi ٦nu, ire, ints enacted after the date of the Agreement affecting the the effects of any changes in Legai performance of the Work, or it aranteeu num Price is established after the date of the Agreement, the date t' arties agree upo <sup>2</sup> Guaranteed Maximum Price. Such effects may include, without uired to make to the Construction Documents because of changes limitation, rign-Builder is in Legal Req ments.

#### 2.6 Government A ov and Permits.

- **2.6.1** Except a l'entified in an Owner's Permit List attached as an exhibit to the Agreement, Design-Builder chall obtained pay for all necessary permits, approvals, licenses, government charges and incompanies are a for the prosecution of the Work by any government or quasi-government entity ang jurisdian nover the Project.
- **2.6.2** De '-Builder shall provide reasonable assistance to Owner in obtaining those permits, approvals that are Owner's responsibility.

#### 2.7 Design-Builder's Construction Phase Services.

- **2.7.1** Unless otherwise provided in the Contract Documents to be the responsibility of Owner or a separate contractor, Design-Builder shall provide through itself or Subcontractors the necessary supervision, labor, inspection, testing, start-up, material, equipment, machinery, temporary utilities and other temporary facilities to permit Design-Builder to complete construction of the Project consistent with the Contract Documents.
- 2.7.2 Design-Builder shall perform all construction activities efficiently and with the requisite expertise,

## REQUEST FOR QUALIFICATIONS (RFQ)

skill and competence to satisfy the requirements of the Contract Documents. Design-

Builder shall at all times exercise complete and exclusive control over the means, methods, sequences and techniques of construction.

- **2.7.3** Design-Builder shall employ only Subcontractors who are duly licensed and qualified to perform the Work consistent with the Contract Documents. Owner may reasonably object to Design-Builder's selection of any Subcontractor, provided that the Contract Price and/or Contract Time(s) shall be adjusted to the extent that Owner's decision impacts Design-Builder's cost and/or time of performance.
- **2.7.4** Design-Builder assumes responsibility to Owner for the proper perform the of the Work of Subcontractors and any acts and omissions in connection with such performance to othing in the Contract Documents is intended or deemed to create any legal or contractual relations between Osmand any Subcontractor or Sub-Subcontractor, including but not limited to any third-part in the Contract performance to other contractor.
- 2.7.5 Design-Builder shall coordinate the activities of all Subcontractors. If Owner forms other k on the Project or at the Site with separate contractors under Owner's control, Design ilder are as to reasonably cooperate and coordinate its activities with thos if such separate contraction at the Project can be completed in an orderly and coordinated manner input unreasonable disrup in.
- Design-Builder shall keep the Site reasonably ash and cor က deb၊ . action wastes to permit Design-Builder to perform its construction vices en. ntly, sa. and v out interfering with the use of adjacent land areas. Upon Substantia' ı of the Work, Designmpletion of : Work, or Builder shall remove all debris, trash, cons rion wastes, terials, equipatent, machinery and tools arising from the Work or applicable portions the f to per Owner to occupy the Project or a portion of the Project for its intended use.

#### 2.8 Design-Builder's Responsibility for Property. V.

- Design-Builder recognizes the in ortan of performing the Work in a safe manner so as to to (i) all indicatuals a site, whether working or visiting, (ii) the Work, prevent damage, injury including materials and incorporated into the Work or stored on-Site or off- Site, and (iii) all יווט: other property at the Site o. 'acen. `to. `sign-Builder assumes responsibility for implementing and monitoring all safety precaution and pros. plated to the performance of the Work. Design-Builder ruction, designate a Safety Representative with the necessary shall, prio commencing cu the implementation and monitoring of all safety precautions and qualificatic a... rience to sup-Unless ...nerwise required by the Contract Documents, Design-Builder's programs re ed to u. Safety Repredictative s' dividual stationed at the Site who may have responsibilities on the . . ety. The Salety Representative shall make routine daily inspections of the Site and Project in add 1 to hold weekl y meetings with Design-Builder's personnel, Subcontractors and others as applicable.
- 2.8.2 Pesign-Bu or and Subcontractors shall comply with all Legal Requirements relating to safety, as we' incressific safety requirements set forth in the Contract Documents, provided that such increspecit requirements do not violate any applicable Legal Requirement. Design-Builder will ort in writing any safety-related injury, loss, damage or accident arising from the Work to Owner's Resentative and, to the extent mandated by Legal Requirements, to all government or quasi-authorities having jurisdiction over safety- related matters involving the Project or the Work.
- 2.8.3 Design-Builder's responsibility for safety under this Section 2.8 is not intended in any way to relieve Subcontractors and Sub-Subcontractors of their own contractual and legal obligations and responsibility for (i) complying with all Legal Requirements, including those related to health and safety matters, and (ii) taking all necessary measures to implement and monitor all safety precautions and programs to guard against injuries, losses, damages or accidents resulting from

their performance of the Work.

#### 2.9 Design-Builder's Warranty.

**2.9.1** Design-Builder warrants to Owner that the construction, including all materials and equipment furnished as part of the construction, shall be new unless otherwise specified in the Contract Documents, of good quality, in conformance with the Contract Documents and free of defects in materials and workmanship. Design-Builder's warranty obligation excludes defects caused by abuse, at ations, or failure to maintain the Work in a commercially reasonable manner. Nothing in this warrants is intended to limit any manufacturer's warranty which provides Owner with greater warranty right chan set for the in this Section 2.9 or the Contract Documents. Design-Builder will provide Own with all materials and equipment furnished as part of the Contract Documents. Design-Builder will provide Own with all materials and equipment furnished as part of the Contract Documents. Design-Builder will provide Own with all materials and workmanship.

#### 2.10 Correction of Defective Work.

- **2.10.1** Design-Builder agrees to correct any Work that is four onto be in conformance we contract Documents, including that part of the Work subject to Section 2. Treof, within a period of our year from the date of Substantial Completion of the Work or any portion of the Cork, or within such longer period to the extent required by any specific warranty included on tract of the work or any portion of the Cork, or within such longer period to the extent required by any specific warranty included on tract.
- 2.10.2 Design-Builder shall, within seven (7) d of receipt o ritten not Jwner that the Work is not in conformance with the Contract Docu its, take mear , ful steps to commence correction of such nonconforming Work, including the correction, noval or placement of the nonconforming Work and any damage caused to other parts of the 'Vork after the nonconforming Work. If Design-Builder fails to commence the necessary steps v ruch sev. () day period, Owner, in addition to any other remedies provided under the Contract of ants, ma avide Design-Builder with written notice that Owner will commence correction of such once rming W with its own forces. If Owner does perform such corrective Work, Design-Builder sha be re asible for all reasonable costs incurred by Owner in performing such corre "the noncon ming k creates an emergency requiring an immediate response, the seven (7) entified rein sha be deemed inapplicable.
- 2.10.3 The one-year period is renced in 2.10.1 above applies only to Design-Builder's obligation to correct reconforming Work at a not intended to constitute a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights or remedic to what a period of limitations for any other rights.

### Article 3

#### **Owner's Services and Responsibilities**

#### **Duty to Cooperate**

- Own shall, inroughout the performance of the Work, cooperate with Design-Builder and perform as responsible so, obligations and services in a timely manner to facilitate Design-Builder's timely and nance of the Work and so as not to delay or interfere with Design-Builder's performance of its obligations and services in a timely manner to facilitate Design-Builder's timely and nance of the Work and so as not to delay or interfere with Design-Builder's performance of its obligations.
- **3.1.2** Owner shall provide timely reviews and approvals of interim design submissions and Construction Documents consistent with the turnaround times set forth in Design-Builder's schedule.
- 3.1.3 Owner shall give Design-Builder timely notice of any Work that Owner notices to be

defective or not in compliance with the Contract Documents.

#### 3.2 Furnishing of Services and Information.

- **3.2.1** Unless expressly stated to the contrary in the Contract Documents, Owner shall provide, at its own cost and expense, for Design-Builder's information and use the following, all of which Design-Builder is entitled to rely upon in performing the Work:
  - **3.2.1.1** Surveys describing the property, boundaries, topography and reforce points for use during construction, including existing service and utility lines;
  - **3.2.1.2** Geotechnical studies describing subsurface conditions, and her sur is describing other latent or concealed physical conditions at the Site;
  - **3.2.1.3** Temporary and permanent easements, zoning and other in remeriand encumbrances affecting land use, or necessary to pr it the proper design and the Project and enable Design-Builder to perform the V.
  - **3.2.1.4** A legal description of the Site;
  - **3.2.1.5** To the extent available, recore awings of are xisting stream at the Site; and
  - **3.2.1.6** To the extent available, envirous notal structure, s, reports and impact statements describing the environmental conditions solutions. Solutions of the extent available, envirous notal structure, s, reports and impact statements describing the environmental conditions of solutions.
- **3.2.2** Owner is responsible for seculing executing necessary agreements with adjacent land or property owners that are necessary to have responsible for all costs, including attorned fees, regret in securing these necessary agreements.

#### 3.3 Financial Information.

- **3.3.1** At Design-Builder's rest, Owned by promptly furnish reasonable evidence satisfactory to Design-Builder's rest, Owned by that Owner has a contract and obligations are contract Documents. If Owner fails to furnish such financial information in a timely manner, Description of the design of the contract Documents and other right permitted under the Contract Documents.
- Designation of the preceding sentence, after execution of the Agreement Designabilities greater than obligations of the Agreement Designabilities greater than selections are existing ligations Designabilities greater than selections of the Agreement Designabilities greater than selections of the Agreement Designabilities greater than selections of the Agreement Designabilities and the Contract Documents.

#### 3.4 Owner's Resentative.

owner's Representative shall be responsible for providing Owner-supplied information and approvals in a timely manner to permit Design-Builder to fulfill its obligations under the Contract Documents. Owner's Representative shall also provide Design-Builder with prompt notice if it observes any failure on the part of Design-Builder to fulfill its contractual obligations, including any errors, omissions or defects in the performance of the Work. Owner's Representative shall communicate regularly with Design-Builder and shall be vested with the authority to act on behalf of Owner.

#### 3.5 Government Approvals and Permits.

- **3.5.1** Owner shall obtain and pay for all necessary permits, approvals, licenses, government charges and inspection fees set forth in the Owner's Permit List attached as an exhibit to the Agreement.
- **3.5.2** Owner shall provide reasonable assistance to Design-Builder in obtaining those permits, approvals and licenses that are Design-Builder's responsibility.

#### 3.6 Owner's Separate Contractors.

**3.6.1** Owner is responsible for all work performed on the Project or at the seby separate contractually require its separate contraction to coor see with, and coordinate their activities so as not to interfere with, Design-Builder in order to sesign-Builder timely complete the Work consistent with the Contract Documents.

#### **Article 4**

#### Hazardous Conditions and Diff Site Co. Stions

#### 4.1 Hazardous Conditions.

- **4.1.1** Unless otherwise expressly provided . Contration ocuments to be part of the Work, Design-Builder is not responsible for any Hazar lous Contrations ountered at the Site. Upon encountering any Hazardous Conditions, Design-Builde atop Wo. mediately in the affected area and duly notify Owner and, if required by Legal Return and, and arment or quasi-government entities with jurisdiction over the Project or Site.
- ected Hazardous Conditions, Owner shall take the 4.1.2 Upon receiving of the presure of s nsure the the Ha dous Conditions are remediated or rendered necessary measures re "inc de Owner retaining qualified independent experts to (i) harmless. Such necessary rsures actually been encountered, and, if they have been ascertain whether Hazardou. ndition Hial measures that Owner must take either to remove the Hazardous encountere (ii) prescribe the re Conditions the Hazardous ditions harmless.
- **4.1.3** Designated to resume Work at the affected area of the Project only after Owner's experiment oviring with will encertification that (i) the Hazardous Conditions have been removed and (ii) all necessary approvals have been obtained from all government and quasi-government entity having jurisdiction over the Project or Site.
- 4.1 'n-Buil will be entitled, in accordance with these General Conditions of Contract, to an Contract Price and/or Contract Time(s) to the extent Design-Builder's cost and/or time of performance //e been adversely impacted by the presence of Hazardous Conditions.
- 1.5 The fullest extent permitted by law, Owner shall indemnify, defend and hold harmless Design-Lesign Consultants, Subcontractors, anyone employed directly or indirectly by any of them, and their officers, directors, employees and agents, from and against any and all claims, losses, damages, liabilities and expenses, including attorneys' fees and expenses, arising out of or resulting from the presence, removal or remediation of Hazardous Conditions at the Site.
- **4.1.6** Notwithstanding the preceding provisions of this Section 4.1, Owner is not responsible for Hazardous Conditions introduced to the Site by Design-Builder, Subcontractors or anyone for

whose acts they may be liable. To the fullest extent permitted by law, Design-Builder shall indemnify, defend and hold harmless Owner and Owner's officers, directors, employees and agents from and against all claims, losses, damages, liabilities and expenses, including attorneys' fees and expenses, arising out of or resulting from those Hazardous Conditions introduced to the Site by Design-Builder, Subcontractors or anyone for whose acts they may be liable.

#### 4.2 Differing Site Conditions.

- 4.2.1 Concealed or latent physical conditions or subsurface conditions at the Site (i) materially differ from the conditions indicated in the Contract Documents or (ii) are of an unusual (iii), are, differing naterially from the conditions ordinarily encountered and generally recognized as werent in the vork are collectively referred to herein as "Differing Site Conditions." If Design-Builder (number of particular of p
- **4.2.2** Upon encountering a Differing Site Condition, Design-bar shall provide prompt watter notice to Owner of such condition, which notice shall not be later than four in (14) days after such condition has been encountered. Design-Builder shall, to the extentially post provide such otice before the Differing Site Condition has been substantially distance and or a red.

# Artic 5

#### 5.1 Design-Builder's Insurance Requirement

- **5.1.1** Design-Builder consible for rocuri and maintaining the insurance for the coverage amounts all as set for surance hibit to be Agreement. Coverage shall be secured from insurance companies authorized to be secured in the state in which the Project is located, and with a minimum rating set forth in the project is located.
- **5.1.2** D so iter's insurance is specifically delete any design-build or similar exclusions that could compromise inverse of the light insurance is specifically delete any design-build or similar exclusions that could compromise inverse of the light insurance.
- 5.1.3 Prior com ucing any construction services hereunder, Design-Builder shall provide Owner in certificates and in effect and will remain in effect for the duration required by the Contract Documents and (ii) no insurance coverage and in effect in the duration required by the Contract Documents and (ii) no insurance coverage are required by the Contract Documents and (ii) no insurance coverage are required to remain all be canceled, renewal refused, or materially changed unless at least thirty (30) days prior to owner. If any of the foregoing insurance coverages are required to remain all payment are reasonably available, an additional certificate evidencing continuation of such coverage and be submitted with the Final Application for Payment. If any information concerning reduction of everage is not furnished by the insurer, it shall be furnished by the Design-Builder with reasonable.

#### 5.2 Owner's Liability Insurance.

**5.2.1** Owner shall procure and maintain from insurance companies authorized to do business in the state in which the Project is located such liability insurance as set forth in the Insurance Exhibit to the Agreement to protect Owner from claims which may arise from the performance of Owner's obligations under the Contract Documents or Owner's conduct during the course of the Project.

#### 5.3 Owner's Property Insurance.

- Unless otherwise provided in the Contract Documents, Owner shall procure and maintain from insurance companies authorized to do business in the state in which the Project is located property insurance upon the entire Project to the full insurable value of the Project, including professional fees, overtime premiums and all other expenses incurred to replace or repair the insured property. The property insurance obtained by Owner shall be the broadest coverage commercially available, and shall include as additional insureds the interests of Owner, Design-Builder, Design Consultants and Sv ntractors of any tier. Such insurance shall include but not be limited to the perils of fire and extended coverage, theft, vandalism, malicious mischief, collapse, flood, earthquake, debris removal and ot perils or cares of loss as called for in the Contract Documents. The property insurance shall include r = n cal loss or dWork, including materials and equipment in transit, at the Site or at another is on as r ມe indicated in Design-Builder's Application for Payment and approved by Owner. The Own onsible for the payment of any deductibles under the insurance required by this Section 5.3.1.
- 5.3.2 Unless the Contract Documents provide otherwise, Contract Procure and main the machinery insurance that will include the interests of Owner, the main the machinery insurance that will include the interests of Owner, the main that will include the interests of Owner, the main that will be made and the insurance required by this Section 5.3.2.
- 5.3.3 Prior to Design-Builder commencing by Work, ( her shall . Design-Builder with 'ce obligation equired by the contract Documents are in certificates evidencing that (i) all Owner's ins full force and in effect and will remain in effect Design Ider has completed all of the Work and has received final payment from Owner ? (ii) no in. overage will be canceled, renewal refused, or materially changed unless at least thi. `days pri ritten notice is given to Design-Builder. Owner's property insurance shall not lapse or \ c. led if Ov. occupies a portion of the Work pursuant to Section 6.6.3 hereof. Owner shall prov ? D. h the necessary endorsements from the า-Buildei insurance company prior to occupying a polion con two Work.
- 5.3.4 ner's preerty instance shall be adjusted with Owner and Design-Any loss covere str 'ees for the insureds as their interests may appear, subject Builder and made payable to th of u. to any applicable mortgage claroceeds received as a result of any loss will be placed in All insure accordance with such agreement as the interested parties may reach. a separate unt and distribute. Any disagre ne. ution of any proceeds will be resolved in accordance with Article rerning the di. 10 hereof.
- 5.3.5 Owne and all of them, all damages covered by property insurance. Design-Ruilder and view against each other and Owner's separate contractors, Design rovided herein, except such rights as they may have to the proceeds of such insurance. View and the proceeds of such insurance. The serence of them to include the root of the

#### 5.4 Bo. and Other Performance Security.

- **5.4.1** If Owner requires Design-Builder to obtain performance and labor and material payment bonds, or other forms of performance security, the amount, form and other conditions of such security shall be as set forth in the Agreement.
- **5.4.2** All bonds furnished by Design-Builder shall be in a form satisfactory to Owner. The surety shall be a company qualified and registered to conduct business in the state in which the Project is located.

# Article 6 Payment

#### 6.0 Schedule of Values.

- **6.0.1** Unless required by the Owner upon execution of this Agreement, within ten (10) 'ys of execution of the Agreement, Design-Builder shall submit for Owner's review and approval a schrouse of values for all of the Work. The Schedule of Values will (i) subdivide the Work into its respective so, (ii) include values for all items comprising the Work and (iii) serve as the basis for monthly progress ments made. Design-Builder throughout the Work.
- 6.0.2 The Owner will timely review and approve the schedule of values so as now y the submission of the Design-Builder's first application for payment. The Owner and Design-Builder's mely resolved; and differences so as not to delay the Design-Builder's submission of its first application for payment.

#### 6.1 Monthly Progress Payments.

- On or before the date established in the ilder shall mit for Owner's Desig. review and approval its Application for Payment r rmed as of the date esting pa ent for a. rk r of the Application for Payment. The Applica nied by all supporting for Paymer hall be acc documentation required by the Contract Doc. nts and/or e alished at the meeting required by Section 2.1.4 hereof.
- **6.1.2** The Application for Paymer arequest ment for equipment and materials not yet incorporated into the Project, provided has Owner is isfied that the equipment and materials are suitably stored at either the Site or ancher and haterials are protected by suitable instrance and (iii) up no pay to the owner will receive the equipment and materials free and clear of all lies.
- 6.1.3 All discounts offere 'Subcontractors and suppliers to Design-Builder for early payment shall accrue one 'dred per Co Design-Builder to the extent Design-Builder advances payment.' Owner advances ment to Design-Builder specifically to receive the discount, Design-Builder ma, Cluc Application. 'ayment the full undiscounted cost of the item for which payment is sought.
- 6.1.4 The A ic in for Payment shall constitute Design-Builder's representation that the Work ibed herein poeen performed consistent with the Contract Documents, has progressed to the point indicated in the A ication for Payment, and that title to all Work will pass to Owner free and clear of all claims indicated in the A ication for Payment, and that title to all Work will pass to Owner free and clear of all claims indicated in the A ication for Payment, and that title to all Work will pass to Owner free and clear of all claims indicated in the A ication for Payment, and that title to all Work will pass to Owner free and clear of all claims indicated in the A ication for Payment, whichever occurs earlier.

#### 6.2 Withholding Jayments.

- or before the date established in the Agreement, Owner shall pay Design-Builder all amounts properly due. If Owner determines that Design-Builder is not entitled to all or part of an Application for Payment as a result of Design-Builder's failure to meet its obligations hereunder, it will notify Design-Builder in writing at least five (5) days prior to the date payment is due. The notice shall indicate the specific amounts Owner intends to withhold, the reasons and contractual basis for the withholding, and the specific measures Design-Builder must take to rectify Owner's concerns. Design-Builder and Owner will attempt to resolve Owner's concerns prior to the date payment is due. If the parties cannot resolve such concerns, Design-Builder may pursue its rights under the Contract Documents, including those under Article 10 hereof.
- 6.2.2 Notwithstanding anything to the contrary in the Contract Documents, Owner shall pay

Design-Builder all undisputed amounts in an Application for Payment within the times required by the Agreement.

#### 6.3 Right to Stop Work and Interest.

**6.3.1** If Owner fails to pay timely Design-Builder any amount that becomes due, Design-Builder, in addition to all other remedies provided in the Contract Documents, may stop Work pursuant to Section 11.3 hereof. All payments due and unpaid shall bear interest at the rate set forth in the prement.

#### 6.4 Design-Builder's Payment Obligations.

**6.4.1** Design-Builder will pay Design Consultants and Subcontractors, in account of work. Design-Builder will impose similar requirements on Design Consultants and Contractors to your those parties with whom they have contracted. Design-Builder will indemnify and detail of the contractors to your anstany claims for payment and mechanic's liens as set forth in Second 7.3 hereof.

#### 6.5 Substantial Completion.

- Design-Builder shall notify Owner when اeves د Work, L the e it permitted in the Contract Documents, a portion of the Work, ⊿ostantially nplete. W. ع (5) days of Owner's sign-Builder receipt of Design-Builder's notice, Owner an jointly inspecd uch Work to verify that it is Substantially Complete in accordance with the ruiremr of the Contract Documents. If such Work is Substantially Complete, Owner shall r are and i. tificate of Substantial Completion that will set forth (i) the date of Substantial Compl f the Wo portion thereof, (ii) the remaining items of Work that have to be completed before final 'y. t, (iii) pru ons (to the extent not already provided in the Contract Documents) establishing Owner and esign-Bu. 's responsibility for the Project's security, maintenance, utilities and insurance pence 3 fine vment, and (iv) an acknowledgment that warranties commence to run on the except as may otherwise be noted in the Certificate f Substantial mpleti of Substantial Completic
- **6.5.2** Upon Substantial Contion of the Work, or, if applicable, any portion of the Work, Owner shall release the period of the Work or completed portion of the Work or completed portion of the Work or completed portion of the Work or complete items of work as not the continuous the of Substitution.
- 6.5.3 Owne tite clon, may use a portion of the Work which has been determined to be Substantially nlete, proving nowever, that (i) a Certificate of Substantial Completion has been issued for the portion of Work a ressing the items set forth in Section 6.6.1 above, (ii) Design-Builder and Owner have obtained the control of their sureties and insurers, and to the extent applicable, the appropriate thorical having jurisdiction over the Project, and

owner an esign-Builder agree that Owner's use or occupancy will not interfere with Design-Builder's completion of eremaining Work.

#### 6.6 Phal Par .it.

- **6.6.1** After receipt of a Final Application for Payment from Design-Builder, Owner shall make final payment by the time required in the Agreement, provided that Design-Builder has achieved Final Completion.
- **6.6.2** At the time of submission of its Final Application for Payment, Design-Builder shall provide the following information:
  - 6.6.2.1 An affidavit that there are no claims, obligations or liens outstanding or

unsatisfied for labor, services, material, equipment, taxes or other items performed, furnished or incurred for or in connection with the Work which will in any way affect Owner's interests;

- **6.6.2.2** A general release executed by Design-Builder waiving, upon receipt of final payment by Design-Builder, all claims, except those claims previously made in writing to Owner and remaining unsettled at the time of final payment;
- **6.6.2.3** Consent of Design-Builder's surety, if any, to final payment;
- **6.6.2.4** All operating manuals, warranties and other deliverables r ared by the Contract Documents; and
- **6.6.2.5** Certificates of insurance confirming that required coverage. "I" main in effectonsistent with the requirements of the Contract Documents.
- **6.6.3** Upon making final payment, Owner waives all claims inst Design-Builder excepted elating to (i) Design-Builder's failure to satisfy its payment obligations, in the failure affects Owner's derests, (ii) Design-Builder's failure to complete the Work consistent with the tract Documents, including defects appearing after Substantial Completion and (iii) the terms of special transfer of the contract Documents.
- 6.6.4 Deficiencies in the Work discovered rSubstantial / ipletion, where ar or not such deficiencies would have been included on the Punch List it rvered er, shall be deemed warranty Work. Such deficiencies shall be corrected by Desi Builder L cions 2.9 and 2.10 herein, and shall not be a reason hhold fin. ryment from Design-Builder, provided, however, that Owner shall be entitled to withhold ع Final Pد nt the reasonable value of completion of such deficient work until such work is comple. \dagger.

# A<sub>1</sub> 'cle 7

## 7.1 Patent and Infringement.

- 7.1.1 Desi, Builder s' any action or proceeding brought against Owner based on any claim any art thereo, or the operation or use of the Work or any part thereof, constitutes that the Work nited States patent or copyright, now or hereafter issued. Owner shall give prompt ingement of whiten notice to sign-Builder of any such action or proceeding and will reasonably provide authority, tance in the defense of same. Design- Builder shall indemnify and hold harmless information and a t all damages and costs, including but not limited to attorneys' fees and expenses ued aga. Owner or Design-Builder in any such action or proceeding. Design-Builder agrees to keep wner inform of all developments in the defense of such actions.
- 7 1.2 In or is enjoined from the operation or use of the Work, or any part thereof, as the result of a concording or copyright suit, claim, or proceeding, Design-Builder shall at its sole expense take reasonable steps to procure the right to operate or use the Work. If Design-Builder cannot so procure such right within a reasonable time, Design-Builder shall promptly, at Design-Builder's option and at Design-Builder's expense, (i) modify the Work so as to avoid infringement of any such patent or copyright or (ii) replace said Work with Work that does not infringe or violate any such patent or copyright.
- **7.1.3** Sections 7.1.1 and 7.1.2 above shall not be applicable to any suit, claim or proceeding based on infringement or violation of a patent or copyright (i) relating solely to a particular process

or product of a particular manufacturer specified by Owner and not offered or recommended by Design-Builder to Owner or (ii) arising from modifications to the Work by Owner or its agents after acceptance of the Work. If the suit, claim or proceeding is based upon events set forth in the preceding sentence, Owner shall defend, indemnify and hold harmless Design-Builder to the same extent Design-Builder is obligated to defend, indemnify and hold harmless Owner in Section 7.1.1 above.

7.1.4 The obligations set forth in this Section 7.1 shall constitute the sole agreement be veen the parties relating to liability for infringement of violation of any patent or copyright.

#### 7.2 Tax Claim Indemnification.

**7.2.1** If, in accordance with Owner's direction, an exemption for all or parchibe Work claimed for taxes, Owner shall indemnify, defend and hold harmless Design-Builder from a strans liability penalty, interest, fine, tax assessment, attorneys' fees or other expenses or costs incurby Design-Builder are sare result of any action taken by Design-Builder in accordance with Owner's directive. The shall make the property of the Work of th

## 7.3 Payment Claim Indemnification.

7.3.1 Provided that Owner is not in breach s contractu bligation z payments to Design-Builder for the Work, Design-Builder shall in nify, defend 7 hold harmles. Jwner from any claims or mechanic's liens brought against Owner or again he Proir as a result of the failure of Design-Builder, or those for whose acts it is responsible nay for a es, materials, labor, equipment, taxes or other items or obligations furnished or inc ັດr or in rection with the Work. Within three (3) days of claim or hanic's lien has been filed, Design-Builder shall receiving written notice from Owner the 'uc commence to take the steps necessary to isco. said cla. r lien, including, if necessary, the furnishing of a mechanic's lien bond. If Design-Builde fails so, Owner will have the right to discharge the claim or lien and hold Design liable for co. and e. ses incurred, including attorneys' fees.

# 7.4 Design-Builder's General In nifica.

- **7.4.1** Dr Ruilder, to the full extent permitted by law, shall indemnify, hold harmless and defend Owner, its fice. Sectors, and engages from and against claims, losses, damages, liabilities, including attorneys' for and engages for bodi. Injury, sickness or death, and property damage or destruction (other than to a Work' in, sectors, anyone employed directly or indirectly by any of them or anyone for a acts any sectors. In may be liable.
- 7.4.2 If an employee of Design-Builder, Design Consultants, Subcontractors, anyone employed directly or interest, distance in the property of the property of

#### 7.5 Owner's General Indemnification.

**7.5.1** Owner, to the fullest extent permitted by law, shall indemnify, hold harmless and defend Design-Builder and any of Design-Builder's officers, directors, and employees, from and against claims, losses, damages, liabilities, including attorneys' fees and expenses, for bodily injury, sickness or death, and property damage or destruction (other than to the Work itself) to the extent resulting from the negligent acts or omissions of Owner's separate contractors or anyone for

whose acts any of them may be liable.

# Article 8

#### Time

- 8.1 Obligation to Achieve the Contract Times.
  - **8.1.1** Design-Builder agrees that it will commence performance of the Work achieve the Contract Time(s) in accordance with Article 5 of the Agreement.
- 8.2 Delays to the Work.
  - **8.2.1** If Design-Builder is delayed in the performance of the Work due to acts, om. as, cor ons, events, or circumstances beyond its control and due to no fault its own or those for whom a suilder is responsible, the Contract Time(s) for performance shall be reathly extended by Change the arrow of example, events that will entitle Design-Builder to an extension the Contract Time(s) include acts or omissions of Owner or anyone under Owner's contract thing set to contract the Contract of Changes in the Work, Differing Site Conditions, Hazardous Conditions, and Form Majeurations.
  - **8.2.2** In addition to Design-Builder's right time extensi or those ever, set forth in Section 8.2.1 above, Design-Builder shall also be entitled to appropri adjustment of the Contract Price provided, however, that the Contract Price shall be adjusted in the Agreement.

# r<u>tic.</u> to the intract ce and Time

- 9.1 Change Orders.
  - **9.1.1** A consider is a written trument issued after execution of the Agreement signed by Owner and Design-lider, so their agree. It upon all of the following:
    - **9.1.1**. The pe of the lange in the Work;
    - 9.1.1.2 amount of the adjustment to the Contract Price; and
      - 3 Th. vtent of the adjustment to the Contract Time(s).
  - **3.1.2** All clages in the Work authorized by applicable Change Order shall be performed under the applicable contract Documents. Owner and Design-Builder shall negotiate in good faith and as evaluations of the Contract Documents of the Contract Documents. Owner and Design-Builder shall negotiate in good faith and as evaluations of the Contract Documents.
  - **9.1.3** If Owner requests a proposal for a change in the Work from Design-Builder and subsequently elects not to proceed with the change, a Change Order shall be issued to reimburse Design-Builder for reasonable costs incurred for estimating services, design services and services involved in the preparation of proposed revisions to the Contract Documents.

#### 9.2 Work Change Directives.

- **9.2.1** A Work Change Directive is a written order prepared and signed by Owner directing a change in the Work prior to agreement on an adjustment in the Contract Price and/or the Contract Time(s).
- **9.2.2** Owner and Design-Builder shall negotiate in good faith and as expeditiously as possible the appropriate adjustments for the Work Change Directive. Upon reaching an agreement 'the parties shall prepare and execute an appropriate Change Order reflecting the terms of the agreement'

#### 9.3 Minor Changes in the Work.

**9.3.1** Minor changes in the Work do not involve an adjustment in the Col. † Price for Contract Time(s) and do not materially and adversely affect the Work, including the design, the arformance ar workmanship required by the Contract Documents. Design-Builder may make minor finges in the hock consistent with the intent of the Contract Documents, provided, however, that the hold shall promptly inform Owner, in writing, of any such changes ar freecord such changes on the uniments maintained by Design-Builder.

## 9.4 Contract Price Adjustments.

- **9.4.1** The increase or decrease in Contract Princesulting from change in the shall be determined by one or more of the following methods:
  - **9.4.1.1** Unit prices set forth he Agree. subsequently agreed to between the parties;
  - **9.4.1.2** A mutually accepte lu sum, p 'rly itemized and supported by sufficient substantiating data to permit evaluation volume.
  - **9.4.1.3** Costs any other, rkups forth in the Agreement; or
  - **9.4.1.4** If an increasor declarate be agreed to as set forth in items 9.4.1.1 through 9.4.1.3 above and Owner iss. A Work of Directive, the cost of the change of the Work shall be declared by the reason the expense and savings in the performance of the Work resulting from the management of the work and profit, as may be set forth in the Agreement.
- **9.4.2** If uncrices are the Contract Documents or are subsequently agreed to by the parties, but application suranit prices will cause substantial inequity to Owner or Design-Builder because of rences in the factor or quantity of such unit items as originally contemplated, such unit prices shall be equitably adjust.
- er a Design-Builder disagree upon whether Design-Builder is entitled to be paid for any sinces required by Owner, or if there are any other disagreements over the scope of Work or proposed changes to the Work, Owner and Design-Builder shall resolve the disagreement pursuant to Article 10 hereof. As report of the negotiation process, Design-Builder shall furnish Owner with a good faith estimate of the perform the disputed services in accordance with Owner's interpretations. If the parties are used agree and Owner expects Design-Builder to perform the services in accordance with Owner's interpretations, Design-Builder shall proceed to perform the disputed services, conditioned upon Owner issuing a written order to Design-Builder (i) directing Design-Builder to proceed and (ii) specifying Owner's interpretation of the services that are to be performed. If this occurs, Design-Builder shall be entitled to submit in its Applications for Payment an amount equal to fifty percent (50%) of its reasonable estimated direct cost to perform the services, and Owner agrees to pay such amounts, with the express understanding that (i) such payment by Owner does not prejudice

Owner's right to argue that it has no responsibility to pay for such services and (ii) receipt of such payment by Design-Builder does not prejudice Design-Builder's right to seek full payment of the disputed services if Owner's order is deemed to be a change to the Work.

## 9.5 Emergencies.

**9.5.1** In any emergency affecting the safety of persons and/or property, Design-Builder shall act, at its discretion, to prevent threatened damage, injury or loss. Any change in the Contract P and/or Contract Time(s) on account of emergency work shall be determined as provided in this Artir.

# **Article 10**

# **Contract Adjustments and Disputes**

## 10.1 Requests for Contract Adjustments and Relief.

**10.1.1** If either Design-Builder or Owner believes that it is entitled elief against the other for any event arising out of or related to the Work or Project, such pr 'provia itten notice t e other party of the basis for its claim for relief. Such notice shall, if incurri, ıble, b د ade pric inv cost or expense d in applications of these General and in accordance with any specific notice requi ents conta quirement, w .en notice shall be given Conditions of Contract. In the absence of a oecific notice within a reasonable time, not to exceed twent, e (21) de after the occurrence giving rise to the claim for relief or after the claiming party real nably should ha. ecognized the event or condition giving rise to the request, whichever is later. Such hall incl ufficient information to advise the other party of lief, the the circumstances giving rise to the cla ific contractual adjustment or relief requested and the basis of such request.

# 10.2 Dispute Avoidance and 'tion.

- **10.2.1** The parties are full pmn. If woring with each other throughout the Project and agree to communicate regularly with each other accommon so as to avoid or minimize disputes or disagreements. If disputes or disagreements do ar. Design-Burder and Owner each commit to resolving such disputes or disagreem a micable, program and expeditious manner so as to avoid unnecessary losses, delays and couption. Work.
- 10.2.2 Design uilder and Owne. All first attempt to resolve disputes or disagreements at the field level etween Design-Builder's Representative and Owner's Representative which shall teen (14) days of the written notice provided for in Section 10.1.1 unless the Owner and Design-Builder attually agree otherwise.
- Just of disagreement cannot be resolved through Design-Builder's Representative and Owner's Representative, Design-Builder's Senior Representative and Owner's Senior Representative, upon the request of the party, shall meet as soon as conveniently possible, but in no case later than thirty (30) of a request is made, to attempt to resolve such dispute or disagreement. Five (5) days prior cetings between the Senior Representatives, the parties will exchange relevant information that will assist the parties in resolving their dispute or disagreement.
- **10.2.4** If after meeting the Senior Representatives determine that the dispute or disagreement cannot be resolved on terms satisfactory to both parties, the parties shall submit within thirty (30) days of the conclusion of the meeting of Senior Representatives the dispute or disagreement to non-binding mediation. The mediation shall be conducted by a mutually agreeable impartial mediator, or if the parties cannot so agree, a mediator designated by the American Arbitration

Association ("AAA") pursuant to its Construction Industry Mediation Rules. The mediation will be governed by and conducted pursuant to a mediation agreement negotiated by the parties or, if the parties cannot so agree, by procedures established by the mediator. Unless otherwise mutually agreed by the Owner and Design-Builder and consistent with the mediator's schedule, the mediation shall commence within ninety (90) days of the submission of the dispute to mediation.

#### 10.3 Arbitration.

- **10.3.1** Any claims, disputes or controversies between the parties arising out or relating to the Agreement, or the breach thereof, which have not been resolved in accordance with the procedures set forth in Section 10.2 above, shall be decided by arbitration in accordance with the procedure of the AAA then in effect, unless the parties mutually agree construction.
- **10.3.2** The award of the arbitrator(s) shall be final and binding upon the parties with the right of are to the courts. Judgment may be entered upon it in accordance with applicable law by violation courts ving jurisdiction thereof.
- **10.3.3** Design-Builder and Owner expressly agree that any arbit. n pursuant to this Section 10.3 may be joined or consolidated with any arbitration ng any ar person or lity (i) necessary to resolve the claim, dispute or controversy, or ' **substan**ւ / involv n or a ced by such claim, dispute or controversy. Both Design-Builder an wner will inc visions in all contracts e appropi they execute with other parties in connectio. th the Project require such juider or consolidation.
- **10.3.4** The prevailing party in any ar' ation, or confinal, binding dispute proceeding upon which the parties may agree, shall be entited the cover of the other party reasonable attorneys' fees and expenses incurred by the prevailing party.

# 10.4 Duty to Continue Performance.

**10.4.1** Unless provided to a trary in the Contact Documents, Design-Builder shall continue to perform the Work and Own hall contact atisfy its payment obligations to Design-Builder, pending the final resolution of any disposit of the contact Documents, Design-Builder shall continue to perform the Work and Own. The contact Documents, Design-Builder shall continue to perform the Work and Own. The contact Documents, Design-Builder shall continue to perform the Work and Own.

#### 10.5 CONSEQUE TA TAGES.

- 10.5.1 NOT THSTANT OF THING HEREIN TO THE CONTRARY (EXCEPT AS SET FORTH IN SECTION 10.5.2 BELOW NEIT OF DESIGN-SUILDER NOR OWNER SHALL BE LIABLE TO THE OTHER FOR ANY SEQUENTIA SES OR DAMAGES, WHETHER ARISING IN CONTRACT, WARRANTY, TORT (INCLUDING NEGLIGENCE), ST. TLIABILITY OR OTHERWISE, INCLUDING BUT NOT LIMITED TO LOSSES OF USE, PROFITS, BUSINESS REPUTA NOR FINANCING.
- the payment liquidated damages limitation set forth in Section 10.5.1 above is not intended to affect the payment liquidated damages or lost early completion bonus, if any, set forth in Article 5 of the Agreement lich both parties recognize has been established, in part, to reimburse Owner or reward are for some damages that might otherwise be deemed to be consequential.

## **Article 11**

## **Stop Work and Termination for Cause**

## 11.1 Owner's Right to Stop Work.

- **11.1.1** Owner may, without cause and for its convenience, order Design-Builder in writing to stop and suspend the Work. Such suspension shall not exceed sixty (60) consecutive days or ap \_\_\_\_\_\_ ate more than ninety (90) days during the duration of the Project.
- **11.1.2** Design-Builder is entitled to seek an adjustment of the Contract Price of Contract of the Contract of the Contract of Stoppe of the Work by Owner.

#### 11.2 Owner's Right to Perform and Terminate for Cause.

- 11.2.1 If Design-Builder persistently fails to (i) provide a suff. the number of skilled work apply the materials required by the Contract Documents, (iii) comply applicable Legal Requirements, (iv) timely pay, without cause, Design Consultants or Subconstructions, (v) acute the Work ith promptness and diligence to ensure that the Work is completed to the Contract of time and such actions and remedies provided in the Contract of the Contract
- **11.2.2** Upon the occurrence of an set forth ction 11.2.1 above, Owner may provide written notice to Design-Builder that it intend. reement unless the problem cited is cured, or ninate th commenced to be cured, within seven ( day Design-L 'er's receipt of such notice. If Design-Builder roblem, an Owner may give a second written notice fails to cure, or reasonably commence to ce, su to Design-Builder of its at to terminal vithin additional seven (7) day period. If Design-Builder, within such second seve ariod, fails cure, c asonably commence to cure, such problem, then Owner may declare the A<sub>b</sub> neni inate or default by providing written notice to Design-Builder of such declaration.
- terminated pursuant to Section 11.2.2 above, Owner may enter ing the Agree inke posses. In for the purpose of completing the Work, of all materials, upon the k mises ances and other items thereon, which have been purchased or provided equipment, s folds, too for the perforunce of the Work of which Design-Builder hereby transfers, assigns and sets over to e, and to employ any person or persons to complete the Work and provide all of the reي ired labor, s ces, materials, equipment and other items. In the event of such termination, Design-Builder shall not by titled to receive any further payments under the Contract Documents until the Work shall." ed in accordance with the Contract Documents. At such time, if the unpaid balance \_contra \_ Price \_xceeds the cost and expense incurred by Owner in completing the Work, such excess Owner to Design-Builder. Notwithstanding the preceding sentence, if the Agreement establishes / Juaranteed Maximum Price, Design-Builder will only be entitled to be paid for Work or to its default. If Owner's cost and expense of completing the Work exceeds the unpaid the Contract Price, then Design-Builder shall be obligated to pay the difference to Owner. Such costs and expense shall include not only the cost of completing the Work, but also losses, damages, costs and expense, including attorneys' fees and expenses, incurred by Owner in connection with the reprocurement and defense of claims arising from Design-Builder's default, subject to the waiver of consequential damages set forth in Section 10.5 hereof.
- **11.2.4** If Owner improperly terminates the Agreement for cause, the termination for cause will be converted to a termination for convenience in accordance with the provisions of Article 8 of the Agreement.

#### 11.3 Design-Builder's Right to Stop Work.

- **11.3.1** Design-Builder may, in addition to any other rights afforded under the Contract Documents or at law, stop the Work for the following reasons:
  - **11.3.1.1** Owner's failure to provide financial assurances as required under Section 3.3 hereof; or
  - **11.3.1.2** Owner's failure to pay amounts properly due under Design-P .er's Application for Payment.
- **11.3.2** Should any of the events set forth in Section 11.3.1 above occur, De. Builder the right to provide Owner with written notice that Design-Builder will stop the Work unless so the seven (7) days from Owner's receipt of Design-Builder's notice. If Owner does not cu. The problem value of the seven (7) day period, Design-Builder may stop the Work. In such case, Design-Builder hall be acled to make a claim for adjustment to the Contract Price and Contract Time(s) to the extension been adversely impacted by such stoppage.

## 11.4 Design-Builder's Right to Terminate for Cause.

- 11.4.1 Design-Builder, in addition to any othe onts and ren ies provio e Contract Documents or by law, may terminate the Agreement for the folling reasons:
  - 11.4.1.1 The Work has be stopped in (o0) consecutive days, or more than ninety (90) days during the duration of t ect, becaula from of court order, any government authority having jurisdiction over the Work, or the work, or the work, or the work, or the work of the stoppages are not due to the actions of the work of
  - 11.4.1.2 Over 's a prove Design Fuilder with any information, permits or approvals that are Owner's answer of the Contract Documents which result in the Work being stopped for sixty (60) secutive are more than ninety (90) days during the duration of the Proventhough Over has not ordered Design-Builder in writing to stop and suspend the Work part to Section 1. Thereof.
  - 11.4 3 O to cure the problems set forth in Section 11.3.1 above after Design-Builde as a ped the Work.
- 11.4.2 Upon the scurrence of an event set forth in Section 11.4.1 above, Design-Builder may provide written notice to C er that it intends to terminate the Agreement unless the problem cited is cured, or be composed by the composed by the

# 11.5 Bankruptcy of Owner or Design-Builder.

**11.5.1** If either Owner or Design-Builder institutes or has instituted against it a case under the United States Bankruptcy Code (such party being referred to as the "Bankrupt Party"), such event may impair or frustrate the Bankrupt Party's ability to perform its obligations under the Contract Documents. Accordingly, should such event occur:

11.5.1.1 The Bankrupt Party, its trustee or other successor, shall furnish, upon request

of the non-Bankrupt Party, adequate assurance of the ability of the Bankrupt Party to perform all future material obligations under the Contract Documents, which assurances shall be provided within ten (10) days after receiving notice of the request; and

**11.5.1.2** The Bankrupt Party shall file an appropriate action within the bankruptcy court to seek assumption or rejection of the Agreement within sixty (60) days of the institution of the bankruptcy filing and shall diligently prosecute such action.

If the Bankrupt Party fails to comply with its foregoing obligations, the non-Bankrup atty shall be entitled to request the bankruptcy court to reject the Agreement, declare the Agreement erminated and pursue any other recourse available to the non-Bankrupt Party under this Article 11.

**11.5.2** The rights and remedies under Section 11.5.1 above shall not be deemed in the ability of the non-Bankrupt Party to seek any other rights and remedies provided by the Contractor occuments of law, including its ability to seek relief from any automatic stays under the United States. Truptor the right of Design-Builder to stop Work under any application of these Generations of Contract.

# Artir 12 Elec 1ic Data

#### 12.1 Electronic Data.

**12.1.1** The parties recognize that ct Docu. 3, including drawings, specifications and three-dimensional modeling (such as Buildin, 1) ation Mc 1) and other Work Product may be transmitted among Owner, Design-Builder and oth 3 in octronic. 'ia as an alternative to paper hard copies (collectively "Electronic Data").

## 12.2 Transmission of Electro.

- 12.2.1 Owner and Design-B. It shall as on the software and the format for the transmission of Electronic F in Each party shall responsible for securing the legal rights to access the agreed-upon format, in all necessary, obling appropriately licensed copies of the applicable software or electronic promoter interpret d/or generate the Electronic Data.
- **12.2.2** Neith particulates and, representations or warranties to the other with respect to the respect to the
- 12.2.3 By transmitting Work Product in electronic form, the transmitting party does not transfer or assign its rights in Work Product. The rights in the Electronic Data shall be as set forth in Article 4 of the Interest of June 1 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which is a set forth in Article 4 of the Interest of Section 2 which it is a set forth in Article 4 of the Interest

#### 12.3 Electronic Data Protocol.

**12.3.1** The parties acknowledge that Electronic Data may be altered or corrupted, intentionally or otherwise, due to occurrences beyond their reasonable control or knowledge, including but not limited to compatibility issues with user software, manipulation by the recipient, errors in transcription or transmission, machine error, environmental factors, and operator error.

Consequently, the parties understand that there is some level of increased risk in the use of Electronic Data for the communication of design and construction information and, in consideration of this, agree, and shall require their independent contractors, Subcontractors and Design Consultants to agree, to the following protocols, terms and conditions set forth in this Section 12.3.

- **12.3.2** Electronic Data will be transmitted in the format agreed upon in Section 12.2.1 above, including file conventions and document properties, unless prior arrangements are made in advance in writing.
- **12.3.3** The Electronic Data represents the information at a particular point in ' and is subject to change. Therefore, the parties shall agree upon protocols for notification by the for to the recipient of any changes which may thereafter be made to the Electronic Data, which precoles shall also for uress the duty, if any, to update such information, data or other information contained in the electronic Data, which precoles the duty if any, to update such information, data or other information contained in the electronic Data in the electronic
- **12.3.4** The transmitting party specifically disclaims all warranties, expressed or implied industry and limited to, implied warranties of merchantability and fitness for a particular purpose, with to the media transmitting the Electronic Data. However, transmission to electronic Data via electronic Data via electronic means shall not invalidate or negate any duties pursuant to the application and and of care with respect to the creation of the Electronic Data, unless such data is more than the particular purpose, with the shall be altered after the stransmitted to the receiving party, and the transmitting party did particular in such that are received as a shall be altered after the particular purpose, with the shall be altered after the particular purpose, with the shall be altered after the

# Artic 13

#### 13.1 Confidential Information.

13.1.1 Confidential Infanction is define. s info. ion which is determined by the transmitting party to be of a confidential confidentia

## 13.2 Assignment.

- Neithe. Ja-Builder nor Owner shall, without the written consent of the other assign, transfer or Lablet any por an or part of the Work or the obligations required by the Contract Documents.
- 3.3 Sucr
  - **3.3.1** Designation 3uilder and Owner intend that the provisions of the Contract Documents are binding upon the parties, for employees, agents, heirs, successors and assigns.
- 13.4 s Law.
  - **13.4.1** The Agreement and all Contract Documents shall be governed by the laws of the place of the Project, without giving effect to its conflict of law principles.

#### 13.5 Severability.

**13.5.1** If any provision or any part of a provision of the Contract Documents shall be finally determined to be superseded, invalid, illegal, or otherwise unenforceable pursuant to any applicable Legal Requirements, such determination shall not impair or otherwise affect the validity, legality, or enforceability of the remaining provision or parts of the provision of the Contract Documents, which shall remain in full force and effect as if the unenforceable provision or part were deleted.

#### 13.6 No Waiver.

**13.6.1** The failure of either Design-Builder or Owner to insist, in any on more instance, on the performance of any of the obligations required by the other under the Continuous shall not be construed as a waiver or relinquishment of such obligation or right with respect to the efformance.

#### 13.7 Headings.

**13.7.1** The headings used in these General Conditions of Cont. or any other Contract Doment, are for ease of reference only and shall not in any way be construe. limit or alter the meaning of any provision.

#### 13.8 Notice.

13.8.1 Whenever the Contract Documents reconstruction that notice will be deemed to have been validly given if deliver a son to the individual intended to receive such notice, (ii) four (4) days after being a register certified mail, postage prepaid to the address indicated in the Agreement, or (iii) if to so, and by factive, by the time stated in a machine generated confirmation that notice was received at the solution of the individual intended to receive such notice, (ii) four (4) days after being a register of certified mail, postage prepaid to the address indicated in the Agreement, or (iii) if to so, and by factive, by the time stated in a machine generated confirmation that notice was received at the solution of the individual intended to receive such notice, (iii) four (4) days after being a register of certified mail, postage prepaid to the address indicated in the Agreement, or (iii) if to so, and a register of the individual intended to receive such notice, (iii) four (4) days after being a register of certified mail, postage prepaid to the address indicated in the Agreement, or (iii) if to so, and a register of the individual intended to receive such notice, (iii) four (4) days after being a register of certified mail, postage prepaid to the address indicated in the Agreement, or (iii) if to so, and a register of the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iii) for the individual intended to receive such notice, (iiii) for the individual intended to receive such notice, (iiiiii) for

#### 13.9 Amendments.

**13.9.1** The Contract Docuents many he anged, altered, or amended in any way except in writing signed by a duly authorized reportative and party.